

Learning from the Origins

Alexander Yarkin



Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email office@cesifo.de Editor: Clemens Fuest https://www.cesifo.org/en/wp An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com

- from the RePEc website: <u>www.RePEc.org</u>
- from the CESifo website: <u>https://www.cesifo.org/en/wp</u>

Learning from the Origins

Abstract

How do political preferences and voting behaviors respond to information coming from abroad? Focusing on the international migration network, I document that opinion changes at the origins spill over to 1st- and 2nd-generation immigrants abroad. Local diasporas, social media, and family ties to the origins facilitate the transmission, while social integration at destination weakens it. Using the variation in the magnitude, timing, and type of origin-country exposure to the European Refugee Crisis of 2015, I show that salient events trigger learning from the origins. Welcoming asylum policies at the origins decrease opposition to non-Europeans and far-right voting abroad. Transitory refugee flows through the origins send abroad the backlash. Data from Google Trends and Facebook suggests elevated attention to events at the origins and communication with likeminded groups as mechanisms. Similar spillovers following the passage of same-sex marriage laws show the phenomenon generalizes beyond refugee attitudes.

JEL-Codes: O150, Z130, D720, D830, P000, J610, F220.

Keywords: immigration, social networks, spillovers, political attitudes, integration.

Alexander Yarkin Brown University Department of Economics 64 Waterman street Providence / RI / USA alexander_yarkin@brown.edu

August 9, 2023

I'm very grateful to Oded Galor, Stelios Michalopoulos, David Weil, and Pedro Dal Bo for their continued advice on this project. Cevat Aksoy, Dany Bahar, Sam Bazzi, Michelle Brock, Julian Costas-Fernandez, Carl-Johan Dalgaard, Ralph De Haas, Frederic Docquier, Ruben Durante, Christian Dustmann, Raquel Fernandez, Martin Fernandez-Sanchez, Sergei Guriev, Casper Worm Hansen, Jonas Hjort, Matt Jackson, Marc Klemp, Brian Knight, Ro'ee Levy, Erzo F.P. Luttmer, Marco Manacorda, Fabio Mariani, Christian Ochsner, Sebastian Ottinger, Louis Putterman, Matthias Schief, Pablo Selaya, Paul Sharp, Andreas Steinmayr, Marco Tabellini, Andrea Tesei, Gerhard Toews, Joachim Voth, Romain Wacziarg, Max Winkler, David Yang, and Yves Zenou provided many valuable comments and suggestions. This paper has also benefited from seminars at Brown, Harvard (Government), Monash (AYEW), UCL (CReAM), Warwick (CAGE), QMUL, EBRD, LISER, Copenhagen, SDU, Bocconi (PE), NES, HSE, and conferences CefES-2021, RES-2022, ASREC-2022, UPenn International Borders, CESifo Global Economy 2022, 3rd KCL Workshop on Migration, SIOE-2022, Economics of Migration Webinar, 12th OECD Immigration conference, and EEA-ESEM 2023.

1 Introduction

The unprecedented rise of global connectivity and cross-border communications over the last decades provides real-time access to information and opinions from all over the world. How do political attitudes, social norms, and voting behaviors respond to information coming from abroad? Consumption of foreign media, maintaining ties with far-away friends and families, attention to distant events - all these channels can be crucial for the spread of political ideologies and cultural norms from abroad. While the existing literature has documented the role of families, local networks, and local shocks in shaping cultures and behaviors¹, the role of distant networks and events is much less understood.

In this paper, I focus on immigration network as a key driver of cross-border social and political spillovers. Since the share of 1^{st} - and 2^{nd} -generation immigrants in many developed countries has reached more than 25-30%, and many of those have voting rights at destination, the dynamics of political preferences among immigrants and their children is crucial for local opinions, election outcomes, and social cohesion. If immigrants maintain sufficiently high attention to events at the origins (which I document in this paper), changes in political opinions can spill over from origins to immigrants and further to locals. Therefore, I ask whether there is a 'home bias' in the dynamics of political opinions: do immigrants adjust their attitudes and voting behaviors following salient developments at the origins? If so, what are the mechanisms behind such spillover effects?

I address these questions in two steps. First, I present a set of stylized facts about the co-movements of political attitudes between immigrants and their countries of origin. Using data from the European Social Survey and the Eurobarometer, I estimate the effects of *changes* in origin-country attitudes on *changes* in the attitudes of immigrants. I find significant co-movements for many salient issues, including opposition to non-Europeans, EU integration, LGBT rights, and trust in politicians. Importantly, co-movements with the origins are stronger for immigrants less socially integrated into their host communities, as measured by citizenship, language use, and the length of residence.

Second, to advance causality, I zoom into the case of the European Refugee Crisis, one of the most salient events in Europe over the past $decade^2$. There is ample evidence show-

¹Fernández and Fogli (2006, 2009), Luttmer and Singhal (2011), Dohmen et al. (2012), Galor and Özak (2016), Bentzen (2019), Aksoy et al. (2020), Giuliano and Nunn (2020), Agostinelli et al. (2020) show evidence for the role of inter-generational transmission, horizontal learning from local networks, and local shocks.

²According to the Eurobarometer, immigration was a top-1 issue facing the EU from 2014 to 2017. 36% of respondents named immigration as a top-1 issue for their country in 2015. It remained in top-3 until 2018.

ing that massive refugee inflows can significantly affect cultural and political landscapes of receiving countries³. But can the effects of such salient events extend beyond country borders and create political spillovers abroad? While anecdotal evidence abounds⁴, we still lack systematic evidence on this matter. I address this gap and document how salient events activate real-time political spillovers through migration networks.

I use data on the inflows of refugees and changes in asylum recognition policies across European countries in 2014-2016, and estimate how 1^{st} - and 2^{nd} -generation immigrants respond to origin-country events. As long as immigrants do not affect reception policies back at the origins, the origin-country treatment can be seen as plausibly exogenous. Using the DID and event study models, I find that accepting and hosting significant numbers of refugees at the origins decreases the opposition to refugees and non-Europeans among the diaspora abroad (as compared to people of other origins living in the same country and region). In contrast, transitory migration through the origins (e.g., Serbia or Hungary) makes the expats more conservative. These reactions mirror opinion changes at the origins and are mediated by the activation of opinion spillovers when issue salience is high.

To diminish the threat of correlated shocks creating spurious spillover effects (Manski (1993)), I restrict the sample to immigrants living in countries not directly exposed to the Crisis (e.g., Poland or Spain). Reassuringly, the results are not driven by correlated exposures. Additional tests show (i) that results are not driven by changing migrant composition during the Crisis; (ii) no Crisis-induced spillovers for non-salient topics⁵; (iii) no spillovers from non-origin countries; (iv) robustness to alternative measures of exposure, and (v) robustness to group-specific changes in opinions at the level of education, political ideology, or broad region (e.g., Southern Europe) of origin.

Importantly, spillovers of opinions have real consequences in terms of voting for far-right parties. I match political parties from the ESS to the PopuList dataset (Rooduijn et al. (2019)), and show that favorable refugee policies at the origins decrease support for far-right parties among the diaspora. In contrast, large transitory migration through the origins increases support for far-right parties among the diaspora, by as much as 10-11% in 2015.

³Among others, Dustmann et al. (2019), Steinmayr (2021), Dal Bo et al. (2019), and reviews in Alesina and Tabellini (2021) and Noury and Roland (2020).

⁴Prominent examples of such spillovers include the Hungarian PM Viktor Orban influencing the opinions and voting behaviors of the Hungarian diaspora in Europe during the Refugee Crisis. The influence of Russian media on the attitudes of "Russian Germans" is another famous example. Finally, Donald Trump's anti-refugee stance often relied on the reactions and attitudes of Europeans to the Refugee Crisis of 2015.

⁵No spillovers for topics unrelated to refugees and non-Europeans, suggesting that the results are driven by attention spikes at the *topic* \times *origin* \times *time* level, and not by the overall attention to specific countries.

What are the mechanisms behind learning from the origins? As a first step, I show that immigrants pay increasing attention to events that are salient at the origins using data from from Google Trends (GT) on searches in different languages. When salient developments happen at the origins, GT searches in origin-country language spike relative to other language searches on the same subject nearby. For example, in France, searches for "Refugees" in German, but not in Spanish, language spike after Angela Merkel's speech in September 2015 and Cologne assaults in January 2016⁶. When origin-country interest related to refugees picks up, this interest spills over abroad (accounting for locality-specific changes in searches), suggesting a disproportionately high demand for origin-country news during salient events.

When immigrants receive news from the origins, why update own attitudes and voting behaviors following these salient events? Conceptually, there could be two main reasons (see also Bursztyn et al. (2014)): identity/social pressure to comply with changing norms at the origins vs. new information from the origin networks that induces opinion change. In the Appendix, I outline a simple model that illustrates the workings of these two channels. The data shows that immigrants more attached to destination (e.g., citizenship) display weaker updating from the origins, suggesting that identity plays a role. However, I also find evidence for three transmission channels suggesting the importance of cross-border communications.

First, online social media ties: immigrants more tightly connected to the origins via Facebook show stronger co-movement. In contrast, immigrants who consume news via TV receive weaker spillovers. Second, cross-border family ties: immigrants who live at destination together with their parents (thus arguably having fewer ties at the origins) show weaker co-movement with the origins. Finally, immigrants living in regions with larger co-national diasporas show stronger reactions to the origins⁷.

Additional evidence supporting the network communication mechanism comes from the patterns of cross-border network homophily. I match respondents to their origin-country peers most similar in terms of political affiliation, and show that most of the spillover comes from like-minded groups. This homophily effect suggest that re-activation of histories or identities common to everyone from a given origin (Ochsner and Rösel (2017) and Fouka and Voth (2021)) is less likely to be the main explanation here. Together with the importance of online social networks, these results suggest that in the era of cost-less access to like-minded groups located far away, there emerges a fertile ground for cross-border political contagion.

⁶I restrict the sample to search locations not directly exposed to the Crisis. In regions with higher population shares from most affected countries, the spikes in attention are stronger.

⁷The latter is, in principle, consistent with both the communication and the social pressure mechanisms, see also Borjas (1992) and Luttmer and Singhal (2011)

Of course, this does not rule out exposure to like-minded media sources at the origins, which is hard to separate from the influence of like-minded peers.

The 'learning from the origins' phenomenon generalizes both across salient events and across space. First, I show that spillovers from the European Refugee Crisis extend to immigrants located in the US. Second, using the staggered passage of same-sex marriage laws across Europe, I show that these reforms affect LGBT attitudes of the expats abroad.

Contribution to the Literature

This paper contributes to several strands of research. First is the literature on persistence and change of preferences, norms and values. The role of inter-generational and horizontal transmission of norms and values has been explored at least since Bisin and Verdier (2000, 2001) and Galor and Moav (2002). In particular, the *epidemiological approach* used in Fernández and Fogli (2006, 2009), Giuliano (2007), Luttmer and Singhal (2011), Galor and Özak (2016), among others, reveled significant effects of origin-country culture and environment on preferences and behaviors of immigrants. Giavazzi et al. (2019) further show that there is an inter-generational convergence to local norms, while Borjas (1992) shows that 'ethnic capital' can slow down this process⁸. The role of local networks affecting the formation of (risk and trust) preferences was shown in Dohmen et al. (2012) and Ahern et al. (2014). Finally, Bentzen (2019), Autor et al. (2020) and Aksoy et al. (2020), among others, document the importance of local salient events: disasters, trade shocks, and new laws. While the role of vertical transmission, local networks and local shocks is well-explored, this paper shows that distant shocks and opinions matter as strongly. It documents that salient events and associated *changes* in opinions at the origins matter for *changes* in opinions at destination.

Second, this paper contributes to the literature on the diffusion of social norms and preferences between populations. The role of migration in diffusing culture and ideas was shown by Barsbai et al. (2017) in the context of 'democratic remittances' to the origins; by Miho et al. (2020) for gender norms exported by German and Chechen deportations in the USSR; by Ochsner and Roesel (2020) in the historical context of Nazi migrants in Austria; and by Burchardi et al. (2019) in the context of the FDI ties of the US firms. Thus, existing literature mostly focused on long-term exchanges of stable cultural norms and ideas via migration⁹. This paper, in contrast, documents real-time learning over preexisting migration networks triggered by the salient events at the origins. Beach and Hanlon

⁸Within-country movers also show traces of both current and previous locations in their food choices (Atkin (2016)) and brand preferences (Bronnenberg et al. (2012)).

⁹Rapoport et al. (2020) argue that cross-country migration induces convergence in values, mostly due to cultural remittances. Giuliano and Tabellini (2020), however, show that immigrants affect preferences of

(2022) is probably the closest in this regard, as it shows how a sharp fertility decline in the 19th century Britain produced similar change among culturally British households living elsewhere. Tian et al. (2020) shows that social distancing practices during Covid-19 are remitted back to Mexico from Mexicans in the US.

Third, this paper contributes to a growing literature on the political economy of immigration and refugee inflows¹⁰. Direct effects of refugees on local attitudes and voting were explored, among others, in Hangartner et al. (2019), Altındağ and Kaushal (2021), Steinmayr (2021), and Ajzenman et al. (2022). Importantly, Steinmayr (2021) shows that conservative reactions were more prevalent in Austrian municipalities experiencing transitory migration, as opposed to municipalities that actually hosted refugees. Similarly, Lebow et al. (2023) find that reactions to Venezuelan refugees were more positive in places that had repeated contact with refugees. What I add is that salient events of this scale not only affect local populations, but are also exported abroad via international migration networks¹¹.

Finally, this paper relates to the literature on networks, polarization, and media. Golub and Jackson (2012) show that polarization in opinions is exacerbated by homophily - a tendency to connect to more similar others. Empirically, Halberstam and Knight (2016) and Allcott et al. (2020) show that social media increases political polarization, and Levy (2021) uncovers the mechanisms by randomizing the exposure to conservative or liberal news of Facebook¹². It is thus well understood that exposure to like-minded news and social echo chambers can increase polarization. This paper suggests that polarizing events and opinions can spread from abroad because of the homophily of cross-border social ties.

The rest of the paper is organized as follows. Section 2 describes the data and documents several stylized facts about political co-movements between immigrants and their countries of origin. Section 3 zooms into the European Refugee Crisis, and shows that origin-country exposure to the Crisis affects opinions and voting behaviors of the expats. Section 4 analyses the mechanisms behind these effects. Section 5 provides generalizations: (i) spillovers from the European Refugee Crisis to the US, and (ii) similar spillovers from the staggered passage of same-sex marriage laws across Europe. Section 6 concludes.

natives at destination, using the data on historic migration from Europe to the US.

¹⁰A recent review in Alesina and Tabellini (2021) discusses evidence on economic vs. cultural motives behind the anti-immigration backlash, both recently and historically.

¹¹At a local scale, Bratti et al. (2020) and Bredtmann (2022) document spillovers from refugees hosting centers to populist voting in the nearby communities within Italy and Germany, respectively. These findings, however, are unrelated to migration networks, and the mechanisms, generally, remain unclear.

¹²For cross-border exposure to traditional media, DellaVigna et al. (2014), Bursztyn and Cantoni (2016).

2 Data and Stylized Facts

In this section, I document broad regularities about political and cultural spillovers from abroad. I show that across a set of topics, countries, years, and several datasets, there is evidence for co-movements of preferences between immigrants and their countries of origin.

2.1 Data on political preferences and social norms

To measure the dynamics of political preferences and social norms across European countries and among emigrants from these countries, I use data from the European Social Survey (ESS) and, for robustness, the Eurobarometer (EB). The nine rounds of the ESS conducted between 2002 and 2019 allow me to (i) track the evolution of norms and preferences across countries and population groups over time, and (ii) identify first- and second-generation migrants.

The sample of interest: 1st- and 2nd-gen immigrants from European countries

My main sample of interest consists of 1st- and 2nd-gen European immigrants: those who were born in a different European country, or those who were born in the country of residence but whose parent(s) were born abroad. For the 1st-gen immigrants, country of origin is own birth country, and for the 2nd-gens, country of origin is mother's birth country¹³.

Table C1 in the Appendix provides summary statistics for 1st- and 2nd-generation immigrants with European origins and compares this group to 'natives' (respondents and their parents born in the country of residence). Immigrants are more likely to have higher education and to be unemployed, as compared to natives. Among immigrants, 35% are 2nd-gen, and 69% are citizens of their host countries. The average number of years since migration is 29, and only 12% have migrated within the 5 years leading up to the survey (so the sample represents long-stay migrants). Immigrants are less likely to identify with the political right and oppose non-European migration into their host countries, as compared to natives.

Table C2 shows the distribution by country of origin. Naturally, more populous countries have larger diasporas abroad. However, countries with significant recent waves of emigration, e.g., Poland or Romania, have disproportionately large pools of immigrants¹⁴.

¹³Alternative definitions of origins, i.e., based on father's birth country, do not affect the results. Moreover, there are cases where (i) respondents were born abroad, but both parents were born in the current country of residence; or (ii) parents were born in yet another country, different from the respondent's. Removing such cases makes the results stronger. Round 1 of the ESS does not record immigrants' generation of migration, so for the sake of comparability, I restrict the main analysis to rounds 2-9 of the ESS.

¹⁴Out of 41 thousand observations, about 34 thousands have non-missing main outcome variables.

Dynamics of political preferences

To illustrate the idea of co-movement in political preferences between the expats and their origins, I focus on the extent of opposition towards non-Europeans - one of the key outcomes in this paper. The ESS question reads "To what extent do you think [country] should allow people from the poorer countries outside Europe?". The "[country]" stands for respondents' country of residence. The scale is from 1 (allow many) to 4 (allow none).



Figure 1: Dynamics of opposition towards non-Europeans, ESS round 2 to round 9.

Figure 1 illustrates the dynamics of opposition to non-Europeans in four countries and among the expats from these countries living elsewhere. A very clear pattern emerges: while the levels may differ, opinion changes among the expats tends to mimic opinion changes at the origins. Figure B1 confirms a strong pattern of co-movement for many countries with at least 5 rounds of data, and Figure B2 shows similar pattern for LGBT attitudes.

2.2 Estimation of co-movement coefficients

To test more rigorously whether and to what extent immigrants mimic origin-country changes in political opinions, I estimate the following model:

$$Y_{i,o,c,t} = \alpha + \beta_{i,o,t} \cdot Y_{o,t}^{Orig.Av} + \eta X'_{i,o,c,t} + \phi_o + \tau_{c,t} + \varepsilon_{i,o,c,t},$$
(1)

where $Y_{i,o,c,t}$ is a measure of attitudes/behaviors of (1st or 2nd generation) immigrant *i* residing in country *c*, tracing ancestry to country *o*, interviewed in period *t*. $Y_{o,t}^{Orig.Av}$ stands

for the average origin-country attitudes in period t, and $X'_{i,o,c,t}$ is a vector of individual control variables. I cluster standard errors at the origin country level¹⁵.

The theory outlined in the Appendix gives a natural way to think about the key parameter I estimate in such a model, $\beta_{i,o,t}$, which captures the extent of co-movement between an immigrant and his or her origin country. Namely, equation (10) suggests that $\beta_{i,o,t}$ is (i) positive, (ii) increasing in immigrants' attachment to the origins, and (iii) increasing in the amount and precision of information coming from the origins. This section tests the first hypothesis, while Sections 3 - 5 test the other two hypotheses.

Most importantly, model (1) accounts for the origin FEs ϕ_o , so the identification of β comes from the effect of *changes* in political attitudes at the origins on *changes* in political attitudes among the expats. This model addresses two other issues with estimating 'peer effects', discussed extensively at least since Manski (1993). First, I rely on the exogenous peer assignment (country of origin), which limits concerns about peer selection. Second, spurious peer effects can emerge due to common shocks. To address this issue, I include host-country × time FEs, $\tau_{c,t}$, which difference out changes in host-country opinions, media exposure, or economic shocks that can produce correlated changes in opinions.

Some of the immigrants may have decided to migrate because of a growing political or cultural distance they felt with the origins. This pattern of selection would suggest that the estimate of β can be biased towards zero. However, I do not find any evidence that a larger difference between origins and diasporas weakens the co-movement. Moreover, in Section 4, I show that immigrants follow the like-minded groups at the origins, so even when emigration is driven by political/cultural reasons, this does not prevent learning from the origins.

2.3 Co-movements in political attitudes: main regularities

I begin with one of my main variables of interest, opposition towards non-Europeans, and report the estimates of co-movements coefficients β from model (1) in Table 1.

In the baseline specification of column (1) that includes origin, destination, and time FEs, a 1-unit increase in the opposition towards non-Europeans at the origin translates into 0.25 units of increase in opposition among the expats from this origin country. Column (2) adds basic individual controls (age and age squared, gender, years of education, marital status, and employment status) with no change to the estimate. Column (3) adds host country changes in attitudes, confirming that co-movements with the origins still matter after accounting

¹⁵Similar models in Nekoei (2013), Akay et al. (2017), and Albert and Monras (2022) showed how origincountry macroeconomic conditions affect immigrants' labor supply, happiness, and location choice.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
VARIABLES		Oppose non-European immigration						
Oppose non-Eur. immigr. (origin)	0.250***	0.256***	0.211***	0.155**	0.156**	0.178**	0.131**	
	(0.084)	(0.088)	(0.071)	(0.064)	(0.066)	(0.070)	(0.061)	
Oppose non-Eur, immigr, (local)		× ,	0.627***		· /	()		
			(0.063)					
Oppose non-Eur. immigr. (origin) x 2 nd gen.						-0.055		
						(0.045)		
						. ,		
Observations	22,273	21,282	21,282	21,282	21,282	21,282	18,470	
Adjusted R-squared	0.114	0.144	0.150	0.152	0.164	0.164	0.180	
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
ESS round FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Country FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Individual controls	No	Yes	Yes	Yes	Yes	Yes	Yes	
Country x round FEs	No	No	No	Yes	Yes	Yes	Yes	
NUTS FEs	No	No	No	No	Yes	Yes	No	
Group x round FEs	No	No	No	No	No	No	Yes	

Table 1: Co-movements with the origins: opposition towards non-European immigrants

Outcome variable is an individual-level response to a question "to what extent do you think [country] should allow people form the poorer countries outside of Europe?", with answers ranging from 1 (allow many) to 4 (allow none). Main treatment variable is the country-round average value for the same question asked in respondent's country of origin. Column (3) adds local country-round averages. Column (6) adds an interaction term with whether a respondent is a second-generation immigrant. Column (7) allows for separate shocks for Left-Center-Right political groups, and for (non)tertiary educated groups of respondents. Standard errors, clustered at the level of origin countries, are in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

for local, potentially correlated dynamics. Column (4) adds host country-by-round FEs to partial out local shocks common to all immigrants in a given host country. Column (5) adds local regional NUTS FEs, and still delivers a highly significant co-movement estimate. Column (6) shows that co-movements in attitudes are very strong for 1st-generation immigrants, and only slightly weaker for children of immigrants. Finally, column (7) accounts for opinion dynamics among political (left-center-right) and educational (tertiary vs. not) subgroups, showing that the results are not driven by selective migration based on education or political leanings. Overall, these results confirm that opposition towards non-Europeans tends to change similarly between diasporas and their origins.

To test whether this pattern of co-movement between origins and immigrants generalizes to other topics, I re-estimate model (1) for a broader set of political preferences and social norms. The estimates of β coefficients on Figure 2 reveal significant co-movements between origins and immigrants for topics that were salient over the last two decades: (i) support for the EU unification, (ii) opposition towards the LGBT rights, (iii) trust in the EU parliament, (iv) trust in local parliaments. For example, if the public in a given country shows declining institutional trust, immigrants from that country living elsewhere also tend to report lower institutional trust, even after accounting for the host-country changes. Another pattern



Co-movement coefficients for various political and cultural values

Figure 2: Co-movements between origins and immigrants across a range of topics.

emerging from Figure 2 is that there is no significant co-movement for arguably more stable cultural values, such as the importance of traditions, or valuing diversity¹⁶.

It is important to note that emigration can be a result of growing differences in political preferences between those who emigrated and those who stayed. In the Appendix, B3 shows the estimates of country-specific β coefficients for two of the topics analyzed above (opposition to non-Europeans and opposition to LGBT rights), together with the measure of long-term difference between those who left and those who stayed at the origins. While there are large differences between movers and stayers for many countries, these differences in levels do not matter for the extent of co-movement with the origins.

For robustness, I conduct similar analysis with the Eurobarometer (EB) and find identical patterns, as reported in Table D1. Among other things, the EB measures perceptions of issue salience, and I document similar co-movements of issue salience between immigrants and their origins. Moreover, in Section 4.1, I also show that online search behavior displays similar patters.

¹⁶It is statistically harder to detect spillovers for slower-moving traits. Thus, insignificant co-movement coefficients can reflect either (i) lack of immigrants' attention, or (ii) lack of identifying variation (or both).

3 European Refugee Crisis: learning from the origins

In this section, I focus on the European Refugee Crisis of 2015 to show that salient events at the origins activate cross-border spillovers and affect political preferences and voting among the diasporas. As long as the expats do not affect asylum policies back at home, origincountry exposure can be seen as plausibly exogenous. Using the variation in the magnitude, timing, and type of origin-country exposure to the Crisis, I show that it affects (i) opposition to non-Europeans, and (ii) voting for far-right parties abroad. These effects are mediated by the activation of opinion spillovers during the periods of high issue salience.

3.1 Measuring exposure to the Refugee Crisis

My main measures of exposure to the Refugee Crisis are based on the UNHCR data on hosted refugees and asylum recognition policies, and on the International Organization of Migration (IOM) data on transitory migration. Figure 3 shows the timing and magnitude of refugee inflows into Europe along the two main routes. The Central Mediterranean Route via Italy was the first to activate in 2014. The most intense period, however, began in 2015 when more than one million of asylum seekers entered Europe via Turkey and Greece.



Figure 3: Irregular border crossings along the two main routes. Based on Frontex data.



Figure 4: Increase in hosted refugees per capita, from 2013 to 2016. Based on UNHCR data.

Figure B4 in the Appendix shows more detailed country \times time variation in asylum populations per capita and asylum recognition policies (application rejection rates), highlighting the effect of the 2014-2016 Refugee Crisis. Using this data, I calculate changes in per capita refugee populations from the end of 2013 (just before the Crisis) to the end of 2016 (after the main events of the Crisis), and display it on Figure 4. Clearly, many countries hosted close to zero refugees during the Crisis, especially compared to those with the biggest increase in hosted refugees (Turkey, Sweden, Austria, Germany, etc.). Because of this clear division, I code countries as Asylum-treated if the increase is above the European average of approximately 2 hosted refugees per 1000¹⁷. More formally:

$$Asylum_{o} = \begin{cases} 1 & \text{if } \Delta R_{o,2013-2016} \ge \Delta R_{EUR,2013-2016} \\ 0 & \text{if } \Delta R_{o,2013-2016} < \Delta R_{EUR,2013-2016}, \end{cases}$$
(2)

where $R_{o,t}$ is the number of hosted asylum seekers and refugees per capita at the end of year t in country o, and R_{EUR} stands for European average. This measure distinguishes between countries that accepted and hosted refugees during the Crisis and those that did not. In my analysis, I use both the continuous origin-country Asylum increase, and the binary measure.

¹⁷The results are robust to moving the cutoff, e.g., including France or excluding Finland from the category of Asylum-treated. Importantly, the increase in hosted refugees per capita is uncorrelated with pre-Crisis opposition to non-Europeans, see Figure B5 in the Appendix.

The second measure captures asylum reception policies more directly and is based on a drop in asylum rejection rates from before the Crisis (2005-2013) to 2014-2016. Figure 5 shows the distribution of rejection rates drops across countries. A larger drop in rejection rates during Refugee Crisis corresponds to a more welcoming asylum policy. And indeed, Figure B6 in the Appendix shows that countries coded as Asylum-treated display a much larger drop in rejection rates during the Crisis.



Figure 5: Drop in asylum application rejection rates, from 2005-2013 to 2014-2016.

Note than main entry countries, Greece and Italy, did not host so many refugees and did not display especially welcoming asylum policies. One of the reasons is that the majority of asylum seekers were only transiting through on their way to Northern European destinations. As was shown in Steinmayr (2021) and Ajzenman et al. (2022), exposure to transitory migration can generate a stronger backlash as compared to hosting refugees. Thus, one can hypothesize that transitory origin-country exposure can create more negative contagion effects abroad. To capture differences in the predominant mode of exposure to the Crisis (transit vs. stay), I use data from the IOM on the numbers of transit migrants, see Figure B7 in the Appendix for details¹⁸. I code countries as Transit-treated if the number of hosted refugees in 2014-2016 is less than half of the number of asylum seekers who entered the

¹⁸For countries not shown on this Figure, the IOM does not record transit migration, as most irregular migrants apply for asylum in countries like Germany, Sweden, Netherlands, etc.

country in that period (i.e., if among those who entered the country, the majority did not stay). The following countries are thus coded as transit countries: Greece, Croatia, Hungary, Italy, Northern Macedonia, Serbia, and Slovenia¹⁹.

Measures of salience and attention to the Crisis

To measure the salience of Refugee Crisis at the origins, I use two additional sources. One is the Eurobarometer survey, which presents respondents with a large list of social and economic issues and asks to select up to two issues that are most salient for the country at the moment²⁰. I calculate the share of respondents that select immigration as one of the most salient issues at each point in time.

Second, I use media coverage data from the GDELT (the Global Database of Events, Language and Tone, see also Koch et al. (2020)) database to calculate the share of articles that mention refugees. Figure B8 in the Appendix shows the dynamics of immigration salience, media attention, and applications for asylum, confirming that these three measures co-move quite closely, with a peak in the second half of 2015 for most exposed countries²¹.

3.2 The Refugee Crisis: causality and threats to identification

To see whether and how strongly origin-country exposure to the Refugee Crisis affects political preferences and voting among the expats, I employ the Difference-in-Differences (DID) and event study strategies. Namely, I test whether emigrants from origin countries that accepted and hosted refugees change their political attitudes and behaviors differently from people of other origins living nearby. I begin with the following simple 2x2 DID model:

$$Y_{i,o,c,t} = \gamma \cdot Asylum_o \cdot Post_t + \eta X'_{i,o,c,t} + \phi_o + \tau_{c,t} + \varepsilon_{i,o,c,t}, \tag{3}$$

where $Y_{i,o,c,t}$ stands for political preferences of immigrant *i* living in country *c*, originating from country *o*, and interviewed in period *t*. My main outcomes are (i) opposition towards non-European migration, and (ii) voting for far-right parties. *Post*_t is an indicator variable that takes the value of 1 for years 2015 and after (results are robust to using the post-2014 indicator). *Asylum*_o takes the value of 1 for origin countries that hosted significant numbers

¹⁹Transitory migration through Albania, Bosnia, and Montenegro was negligible up until 2017-2018.

²⁰The Eurobarometer question is "What do you think are the two most important issues facing (OUR COUNTRY) at the moment?", and the list includes topics like Economic Situation, Crime, Inflation, Unemployment, Immigration, Terrorism, Housing, Health, and so on.

²¹The exceptions are Greece, Italy, and a few other countries, where as discussed above, applications for asylum were low, but transit migration was high.

of refugees during the Crisis, as defined in 2. An alternative measure, as explained above, is based on the openness of asylum recognition policies, i.e., a drop in asylum rejection rates. Parameter γ captures the main treatment effect: differential change in political preferences between (1st- or 2nd-gen) immigrants coming from Asylum vs. non-Asylum origins.

The identification assumption behind this approach is that origin country exposure to the Crisis is orthogonal to pre-Crisis dynamics of political preferences among the expats, conditional on fixed differences across origins, ϕ_o , and local dynamics, $\tau_{c,t}$. One potential violation of this assumption could happen if diasporas abroad affected reception policies back at the origins, reflecting changing political preferences among the diaspora. Specification with lags and leads explicitly tests for such pre-trends:

$$Y_{i,o,c,t} = \sum_{\tau=2004}^{2019} \gamma_{\tau} \cdot Asylum_o + \eta X'_{i,o,c,t} + \phi_o + \tau_{c,t} + \varepsilon_{i,o,c,t},$$
(4)

I verify that there are no significant pre-trends in political preferences of the expats, so origin-country exposure can be seen as plausibly exogenous for the $expats^{22}$.

To make sure that I capture spillovers from origin to immigrants, and not the other way around²³, I use the fact that immigrants living in host countries that did not experience the Refugee Crisis are more likely to receive, rather than to send, political cross-border spillovers²⁴. This strategy, together with destination × time FEs, $\tau_{c,t}$, also safeguards against the concern of correlated shocks creating spurious spillovers, see Manski (1993).

Additionally, the Refugee Crisis at the origins can induce emigration, changing the composition of migrants abroad and thus creating a spurious effect of origin-country exposure on the average attitudes among the expats. For example, if an inflow of refugees into a country produced backlash towards non-Europeans and made those with the most negative reactions leave the country, then the average opposition towards non-Europeans could have increased both at the origins and among the expats, without any opinion change among non-recent migrants. To address this concern, I omit those migrants who left their origins during or just before the Crisis. The results remain intact.

²²The number of refugees that a given country receives is of course an endogenous outcome that balances the demand for asylum (preference for richer and more tolerant places) with the supply of asylum (e.g., rejection rates, political statements in receiving countries). However, as long as these forces are orthogonal to pre-Crisis dynamics of opinions among the expats, the origin-country exposure to the Crisis is as good as random, accounting for fixed differences across origins (income, tolerance levels, etc.).

²³The paper by Barsbai et al. (2017) has documented 'cultural remittances': immigrants can affect culture and politics at the origins due to exposure to alternative cultures and preferences at destinations.

²⁴For example, one would compare changes in political preferences of a German immigrant (Asylum origin) and a Polish immigrant (not-Asylum origin) living in the same region of Spain (non-affected destination).

Moreover, I test whether the type of origin-country exposure (transit vs. asylum) matters for how the expats react to origin-country events: I substitute $Asylum_{o,t}$ with $Transit_{o,t}$ indicator. In some specifications, I estimate their effects together. Moreover, to make control units more comparable, I test for the effects of asylum/transit exposure relative to their geographical neighbors from control group. One example would be to compare the dynamics of political preferences of immigrants from Hungary (Transit-treated) and those from neighboring Czech Republic and Slovakia (control).

3.3 Main DID and event-study estimates

I begin with a simple 2x2 DID model which shows how hosting refugees at the origins in 2014-2016 affects political preferences abroad. In an event-study model, I show that there are no significant pre-trends, and that spillover effects are the strongest during the Crisis's peak years of 2015-2016. Moreover, the type of exposure matters: transitory migration through the origins has the opposite (negative) effect abroad as compared to Asylum treatment.

When country of origin hosts refugees

Table 2 reports the DID estimates of how origin-country asylum treatment during the Crisis affects the opposition towards non-Europeans among the expats abroad.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
VARIABLES			Oppose no	n-European	immigration	ι	
Origin Asylum-treated x Post2015		-0.134***	-0.131***	-0.117***	-0.095*	-0.168***	-0.110^{***}
Asylum (per capita increase) x Post2015	-0.005* (0.002)	0.000 (0.002)	(0.032)	(0.037)	(0.040)	(0.055)	(0.032)
Observations	32,021	32,021	32,021	28,195	17,911	15,459	31,414
Adjusted R-squared Origin FE	0.126 Yes	0.127 Yes	0.155 Yes	0.159 Yes	0.170 Yes	0.127 Yes	0.155 Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Country x round FEs	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Individual controls	No	No	Yes	Yes	Yes	Yes	Yes
Sample	Full	Full	Full	No transit origins	Controls neighbors	Host not treated	No recent migrants

Table 2: DID estimates of origin-country asylum treatment

Outcome variable is an individual-level response to a question "to what extent do you think [country] should allow people form the poorer countries outside of Europe?" with answers ranging from 1 (allow many) to 4 (allow none). Main treatment variable is the origin-country treatment status interacted with the post-2015 dummy. Treatment definition is based on the increase in hosted refugees per capita from 2013 to 2016. In columns (1) and (2), I also include continuous treatment variable – per capita increase in hosted refugees from 2013 to 2016. Column (3) removes transit migration origins from the control group. Column (4) only retains neighbors of treated origins in the control group. Column (5) only retains the expats currently residing in non-treated host countries. Column (6) only retains the expats currently residing in host countries not neighboring treated countries. Column (7) omits immigrants who left their origins after 2012. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

Column (1) uses the basic specification from model (3) and begins with a continuous origincountry asylum exposure, as per Figure 4. It shows that the more refugees (per capita) an origin country hosts during the Crisis, the lower becomes the opposition towards non-Europeans among the expats abroad. Column (2) shows that the whole effect is driven by the difference between groups of countries - those who hosted virtually zero refugees during the Crisis vs. those who hosted non-negligible numbers of refugees. Column (3) adds individual controls (gender, age, age squared, education, marital status and employment status), and confirms that these variables do not affect the results²⁵. Quantitatively, the estimate from column (3) corresponds to approximately a 14% decrease in reporting a negative attitude towards non-Europeans, relative to the pooled pre-Crisis mean.

In the subsequent specifications, I account for several important identification concerns. Columns (4) and (5) ensure that control units are more comparable to treatment units: first, I remove from the control group those origin countries that had significant transitory migration. Second, I only retain control origins that neighbor treated origins. Results remain almost intact. Column (6) only retains those immigrants who live in host countries that are not themselves exposed to the Crisis (neither hosting refugees, nor having large transitory migration), thereby addressing the issue of potentially correlated exposures to the Crisis. If anything, the effect of the Crisis becomes stronger, suggesting that learning from the origins is more important when there is less local information (consistent with the model in Appendix A). Column (7) addresses the concern that Crisis-induced migration might have changed the composition of expats and thus produced a spurious spillover effect. I drop all post-2013 migrants and show that the results are not driven by selective migration.

To show that there are no significant pre-trends between immigrants from origins to be treated and those not to be treated in the future, I estimate a dynamic DID model (4) and report yearly estimates on Figure 6. The absence of significant pre-Crisis differences between expats from different origins suggests there is no selection into treatment based on characteristics of the expats (origin-country treatment status can be seen as plausibly exogenous). The effect of hosting refugees at the origins is most pronounced in 2015, the peak year of the Refugee Crisis: immigrants from origin countries that accepted and hosted refugees were 0.2 units less negative towards non-Europeans than comparable immigrants

²⁵There are meaningful effects of these characteristics: e.g., better education corresponds to lower opposition to non-Europeans, while unemployment tends to increase it. Importantly, differential changes in opinions among different groups (low vs. high education, or political Left vs. Right) do not explain much of the origin-country treatment effect. Thus, differences in reactions to the Crisis among the expats are not driven by the fact that some origins tend to have more tolerant or more educated expats than others.



Figure 6: Dynamic effects of hosting refugees at the origins on opposition to non-Europeans abroad

from non-Asylum origins. This difference corresponds to approximately 20% lower likelihood of reporting negative attitude towards non-Europeans, relative to a pooled pre-Crisis mean. Moreover, the effect remains very strong for 2016, and still visible up until 2019. It is important to note that this result is robust to allowing separate time FEs for people with Western European origins, suggesting that the effect is driven by refugee policies at the origins, and not to by any factors common to Western European culture.

Table C3 in the Appendix shows that origin-country Asylum treatment affects a broader spectrum of opinions about refugees. Namely, if an origin country accepts and hosts refugees during the Crisis, immigrants from such a country become increasingly supportive of the opinions that (i) local governments should be more generous accepting refugees, (ii) refugees are in real fear of persecution, and (iii) refugees should be entitled to bring close family members²⁶. Moreover, support for income redistribution increases if origins hosts refugees during the Crisis, which is consistent with the recent evidence in Alesina et al. (2018, 2021) on a positive link between attitudes towards migrants and redistribution preferences.Origincountry Asylum treatment also reduces concerns over security issues and perceptions that immigrants worsen crime problems. These additional outcomes confirm that hosting refugees

²⁶I am not using focusing on these outcomes because these questions only available in rounds 1, 7, and 8.

at the origins makes the expats more friendly and less stereotypical towards refugees.

To show that the effect of the Refugee Crisis at the origins is specific to refugee- and migration-related attitudes, I report a series of placebo checks in the Appendix Table C4. Focusing on political and cultural values for which I found significant co-movements with the origins in Section 2, I show that there is no effect of the 2015 Refugee Crisis treatment on these other topics. The exception is a positive effect of Asylum treatment on generalized trust among the expats, which, however, is not robust to stricter dynamic specifications.

Finally, I use an alternative measure of origin-country treatment in the Crisis - the openness of asylum recognition policy during the Crisis, see Figure 5. The hypothesis is that the larger is the drop in asylum rejection rates during the Crisis (2014-2016), as compared to the pre-Crisis period, the more positive spillovers would be sent abroad via migration networks. In the Appendix, Table C5 tests this hypothesis and finds support for it: a larger drop in asylum rejection rates at the origins makes the expats less negative towards non-European migration in their host countries. Overall, there is a strong evidence that more positive asylum treatment at the origins during the period of high issue salience creates positive spillover effects abroad via migration networks.

When origins experience transitory migration

Experience of transitory refugee migration can be very different from actually hosting and interacting with refugees. The contact hypothesis by Allport (1954) suggests that sustained positive contact with out-group members can reduce prejudice. Moreover, a recent paper by Steinmayr (2021) showed that conservative voting increased in Austrian municipalities that experienced transitory migration, but dropped in municipalities that hosted refugees. Does a transitory origin-country exposure to refugees produce negative spillovers abroad?

Figure 7 reports the by-round²⁷ estimates of origin-country transitory migration treatment. It shows that when an origin country is exposed to transitory migration during the Refugee Crisis, the expats from such a country become more hostile to non-Europeans. This result is robust for allowing separate time FEs for people with South European origins²⁸.

²⁷With yearly estimates, most of the treatment effect comes from the year 2014 as opposed to the 2015-2016, potentially because a large part of transitory migration was happening through already in 2014.

²⁸Overall, however, the asylum treatment at the origins has a stronger and more robust effect on the opinions of immigrants. One potential reason behind this is that transitory migration was a very brief experience, while hosting refugees was a more prolonged treatment that did not end in 2015.



Figure 7: Dynamic effect of origin transit treatment on opposition to non-Europeans abroad

3.4 Political spillovers affect behavior: Far-right voting

Documented spillovers of origin-country events and preferences via international migration network constitute an important channel of political and cultural change, and contribute to the contagion of anti-immigration sentiment across the globe. But do these spillovers have real consequences in terms of voting behaviors? In this section, I test whether Refugee Crisis at the origins affects support and voting for far-right parties among the diasporas.

To classify European political parties into far-right vs. not, I rely on a widely used database: the PopuList, by Rooduijn et al. (2019), see also Guriev and Papaioannou (2022). I match parties that the ESS respondents support/voted for in last elections to the PopuList classification, and test whether the origin-country exposure to the Refugee Crisis of 2015 affects support for host-country far-right parties among the emigrants²⁹.

I begin with the outcome variable that gives a party that respondents support at the moment. This variable, while not reflecting actual voting, allows to see an immediate reaction to the origin-country events. Column (1) of Table 3 shows that immigrants from origin countries that accepted and hosted refugees during the Crisis became less likely to support far-right parties at destination, as compared to immigrants from other countries living nearby³⁰. In contrast, as column (2) shows, massive transitory refugee flows through the

²⁹Figure B9 in the Appendix reveals a very strong positive correlation between the opposition towards non-Europeans and support/vote for far-right parties among 1st- and 2nd-generation immigrants.

³⁰The results are very similar for changes in support for populist or far-right populist parties, because in Europe, most populist parties are right-wing, see Rooduijn et al. (2019), Guriev and Papaioannou (2022).

origins significantly increased support for far-right parties abroad. Both types of exposure have an independent and strong effect when estimated together in column (3). Transitory refugee flows at the origins increase support for far-right parties among the diaspora abroad by more than 6 p.p., while favorable asylum policies at the origins decrease far-right support among the diaspora by 5 p.p.

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	Support a f	ar-right poli	itical party	Voted for a far-right political pa		
Origin Asylum-treated x Post2015	-0.045**		-0.049***			
	(0.018)		(0.015)			
Origin Transit-treated x Post2015		0.062***	0.066***			
		(0.018)	(0.014)			
Origin Asylum-treated x Elect. Post2015				-0.047**		-0.048**
				(0.023)		(0.021)
Origin Transit-treated x Elect. Post2015					0.037	0.040*
					(0.024)	(0.020)
Observations	11,343	11,343	11,343	10,099	10,099	10,099
Adjusted R-squared	0.120	0.120	0.121	0.120	0.120	0.121
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes
Country x round FEs	Yes	Yes	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes	Yes

Table 3: DID estimates of spillovers to far-right voting abroad

Outcome variable in columns (1)-(3) is an individual-level political party voting preference from the ESS. Outcome variable in columns (4)-(6) is political party a respondent voted for in the last national elections. Party coding into far-right vs. not is based on the PopuList (Rooduijn et al. (2019)) classification. Both outcomes are $\{0,1\}$ indicators, where 1 stands for a far-right political party in the current country of residence. The sample size is reduced because not all respondents specify a party they support (and even fewer immigrant respondents can vote), and because the merge between the PopuList and the ESS is not perfect. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

Turning to the effects of origin-country events on the actual voting choices of immigrants, it is important to note that for many respondents interviewed in or after 2015, the last election still refers to a pre-Refugee Crisis period. To properly account for this fact, in columns (4)-(6) of Table 3 I compare respondents interviewed before 2015 to those interviewed in or after 2015 and having the last election in or after 2015. In all specifications, the destination \times time changes in far-right voting are differenced out. Column (4) shows that immigrants from countries that accepted and hosted refugees in 2015 were 4.7 p.p. less likely to vote for far-right parties in the post-2015 elections, as compared to other immigrants in the same host country. Columns (5) and (6) show that large transitory refugee shocks at the origins increased far-right voting in the post-2015 elections by 4 p.p.³¹.

³¹While the estimates are smaller and less significant, they are based upon naturalized immigrants who can vote in their host countries. Thus, these are likely to understate the effect of origin-country events on political preferences and (desired) voting of non-naturalized immigrants.



Figure 8: Yearly estimates of origin Transit treatment on far-right voting among the diaspora.

To test for pre-event differences in voting across immigrants of different origins, and to see when the treatment effect is most pronounced, I estimate dynamic DID specifications with lags and leads on Figure 8, using voting for far-right parties in last elections as an outcome. The results reveal that there are no significant pre-trends, and that the effect is most pronounced in 2015 (peak of the Crisis), when there is a 10 p.p. higher chance of voting for a far-right party by immigrants from Transit origins, as compared to immigrants from other countries living nearby. In the Appendix, Figure B10 reports similar estimates for current support, as opposed to past voting, for right-wing parties. The effect on right-wing support remains visible in 2018-2019³².

3.5 Refugee Crisis activates political spillovers from the origins

In this section, I demonstrate that origin-country exposure to the Crisis triggers immigrants' learning from the origins. Namely, political spillovers, identified in section 2 intensify during the periods of high issue salience, and explain why origin-country events affect political preferences and voting among the diasporas abroad.

³²The results are robust to controlling for the dynamic effects of Asylum treatment, including NUTS region FEs, allowing for separate dynamics for groups with different general political ideology (left vs. right) and different education levels (like in Table 1 column (7)).

On Figure 9 I show that political attitudes of the expats respond to origin-country events similarly to how attitudes evolve back at the origins. Asylum-treated countries display a decrease in the opposition to non-Europeans during the Crisis. In contrast, countries that did not accept/host refugees display an increasing opposition to non-Europeans. Similar dynamics is observed for the expats from Asylum-treated vs. other origins. Naturally, spillovers seem to active in 2015 - when the Refugee Crisis becomes most salient.





Figure 9: Dynamics of opposition to non-Europeans, by treatment status (asylum destinations vs. not). Averages after partialling out individual controls, year FEs and country FEs.

Table 4 documents the importance of spillover intensification more rigorously. Column (1) shows that spillovers of opposition to non-Europeans, estimated in Section 2, intensify after 2015. Columns (2) and (3) report the estimates of the basic Asylum and Transit treatment effects, similar to Table 2, but on a smaller sample of respondents, for which origin-country attitudes are available in the ESS. Column (4) then shows that a large part of the Asylum-treatment effect and most of the Transit-treatment effect is accounted for by the intensification of opinion spillovers after 2015. Importantly, Appendix Figure B11 shows no spillover increase after 2015 for topics not directly related to the Refugee Crisis.

Data from the Eurobarometer (EB) gives additional support for the importance of issue salience. In Appendix D, Figures D1 - D4, I show that when the Crisis hits the origins, and immigration becomes a salient issue there, the diasporas living in less affected places also tend to report elevated salience of immigration. All this evidence suggests that the activation of opinion spillovers from origins during the Crisis is the main reason why origin-country events affect preferences and voting abroad.

	(1)	(2)	(3)	(4)			
VARIABLES	Oppose non-European immigration						
Origin Asylum-treated x Post2015		-0.141***	-0.141***	-0.118***			
		(0.037)	(0.032)	(0.028)			
Origin Transit-treated x Post2015			0.119***	0.062			
			(0.039)	(0.054)			
Oppose non-Eur. immigr. (origin)	0.077			0.098*			
	(0.060)			(0.055)			
Oppose non-Eur. immigr. (origin) x Post2015	0.162***			0.066			
	(0.046)			(0.067)			
Observations	20,896	20,896	20,896	20,896			
Adjusted R-squared	0.152	0.152	0.152	0.152			
Origin FE	Yes	Yes	Yes	Yes			
Time FE	Yes	Yes	Yes	Yes			
Country x round FEs	Yes	Yes	Yes	Yes			
Individual controls	Yes	Yes	Yes	Yes			
Sample	ESS Origins	ESS Origins	ESS Origins	ESS Origins			
Outcome maniphia is an individual largel meanance	to a manufiam ()	to milest series	do you think [الم المحمد الم المستخدم م			

Table 4: Political spillovers after the Crisis explain part of the treatment effect

Outcome variable is an individual-level response to a question "to what extent do you think [country] should allow people form the poorer countries outside of Europe?", with answers ranging from 1 (allow many) to 4 (allow none). Columns (1) and (4) include the country-round average value for the same question asked in respondent's country of origin, interacted with a Post-2015 indicator. Columns (2)-(4) add origin-country Asylum- and Transit-treated treatment. The sample is limited to immigrants with origins covered by the ESS. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

Thus far, this paper has established that political preferences and voting behaviors of 1st and 2nd-generation immigrants respond to the Refugee Crisis at the origins. More welcoming asylum policies (accepting and hosting refugees) at the origins decrease opposition to non-Europeans and voting for far-right parties among the diaspora abroad. Massive transitory migration through the origins sends abroad the backlash and an increase in far-right voting. These effects are mediated by the activation of opinion spillovers from origins to immigrants when issue salience is high. This provides new evidence on how political ideology (both conservative and liberal) spreads across the world via pre-existing migration networks. But can we further unpack the mechanisms behind this phenomenon?

4 Mechanisms: communication, attention, and homophily

This section explores the mechanisms behind the observed spillovers of political preferences from origins to immigrants. First, using multiple-language online search data, I document that during the period of high issue salience, immigrants from heavily affected origins pay disproportionately high attention to the Refugee Crisis. Second, I show that network homophily drives most of the effect: the strongest political spillovers come from like-minded groups at the origins. Finally, I underscore the role of social integration at destination and network ties to the origins as important mediators of learning from the origins.

4.1 Google Trends and attention to origin-country events from abroad

Do immigrants pay special attention to salient events at the origins, above and beyond the effects of local salience, and dynamics common to other population groups? The existence of such disproportionate attention is a necessary ingredient for the 'home bias' in political preference dynamics. Traditional data sources, however, do not allow measuring attention to origin-country events. To bypass this issue, I leverage search intensity data from Google Trends (GT). Namely, to measure attention to the Refugee Crisis coming from people outside of their origin countries, I use search volumes for "Refugees" and related terms in different European languages coming from different search locations.

It is important to note several features of the Google Trends data. First, GT provides search intensity relative to the highest point for a given territory and time. For each search term, the GT index varies from 100 (peak search intensity) to 0. To make searches comparable across time and space, a given "search term \times territory \times time" unit needs to remain indexed as a 100 for all search queries. The highest value for foreign-language searches related to the topic of "Refugees" belongs to German-language searches coming from Denmark in September 2015, so this country-language-time is given an index value of 100^{33} . To measure attention to the topic of "Refugees" at the linguistic origins, in local language, all search indices are made relative to the search in Austria in September 2015.

Second, the raw GT index is adjusted for the overall search volume coming from a given territory. Assuming that googling per person is relatively stable across space, one can interpret the GT index in per capita terms. However, the raw GT index is not adjusted for the linguistic composition of a region. Thus, to make foreign language search indices comparable, I account for the share of local (country c) population speaking origin o language, $s_{o,c}$.

Figure 10 illustrates the dynamics of interest in "Refugees" for several European languages, both from within the linguistic homelands and from abroad. This measure clearly captures the growing salience of the Refugee Crisis from the beginning of 2015, with the

³³The value of 50 means that a given country-language-time search for "Refugees" was half as popular as the index one. The GT reports data as missing if the search volume is too low (e.g., too few speakers of a given language in a given residence country). As the result, I end up with 11 languages: BG, DE, ES, FR, HU, IT, PL, RO, RU, SE, TR. I also exclude English searches due to the widespread use of English.



Figure 10: Google Trends search intensity: in selected origin countries (left), and in these origins' languages but coming from abroad (right), population adjusted.

peak of interest in the Fall of 2015. Moreover, the interest in concentrated among Germanspeakers, both at the origins and abroad, with spikes after Angela Merkel's speech in September 2015, and after Cologne assaults in January 2016. On this Figure, and in what follows, I omit searches coming from most affected countries (Turkey, Greece, North Macedonia, Serbia, Italy, Hungary, Austria, Germany, Sweden, and Switzerland), thereby restricting the sample to relatively less affected locations to avoid the issue of correlated exposures.

Spillovers of attention to the Crisis

Does interest in the Refugee Crisis from abroad respond to interest at the origin? For illustration, consider two groups of immigrants living in France and tracing origins to neighboring countries: to Spain (relatively less affected country), and to Germany (heavily affected country). As Figure B12 demonstrates, search volumes for "Refugees" in German ("Flüchtlinge") and in Spanish ("Refugiados") coming from France are no different prior to 2014. However, once the Refugee Crisis begins, German-speakers google about refugees much more actively, which culminates in a 3-fold difference at the peak of the Crisis in September 2015³⁴. An even stronger difference in attention is observed after the Cologne sexual assaults in January 2016: attention of German-speakers spikes, while there is no reaction from Spanish-speakers.

To address this more rigorously, I estimate the following model:

$$Interest Ref_{c,o,t} = \alpha + \beta \cdot Interest Ref_{o,t}^{orig} + \psi_{c,t} + \phi_o + \tau_t + s_{o,c} + \varepsilon_{c,o,t}$$
(5)

where $InterestRef_{c,o,t}$ is interest in the topic of "Refugees" (measured via search volume)

³⁴The month with two of the most salient events of the Crisis: pictures of Alan Kurdi's dead body on the shores of Turkey, and Angela Merkel's "Wir schaffen das" speech, welcoming refugees to Germany.

coming from country c in language o, in month t. Time component τ_t captures how salient the topic is at any given point in time. ϕ_o allows for differences across origins in average interest in the topic (due to, say, distance from the main events or other reasons). InterestRef_{o,t}^{orig} is the measure of origin-country interest. Parameter β captures the extent to which interest in the topic spills over from the origins to abroad. In all specifications I control for $s_{o,c}$ (share of country c population coming from country o), and for (search)country \times month FEs, $\psi_{c,t}$.

Table 5: Spillovers of attention to the Crisis: Google Trends multiple language search data.

	(1)	(2)	(3)
VARIABLES		GT Index	
GT Index (origin)	0.355***	0.215***	0.254***
	(0.033)	(0.059)	(0.028)
GT Index (origin) x Origin population share		55.956***	
		(16.837)	
Observations	17,549	17,549	11,227
Adjusted R-squared	0.533	0.595	0.495
Origin FE	Yes	Yes	Yes
Country FE	Yes	Yes	Yes
Month FE	Yes	Yes	Yes
Country x Month FE	Yes	Yes	Yes
Sample	Full	Full	Host not
			treated

GT Index measures search intensity for term "Refugees" coming from outside of linguistic origins. GT Index (origin) is a corresponding search intensity coming from linguistic origins. Included languages (with sufficient non-missing data): BG, DE, ES, FR, HU, IT, PL, RO, RU, SE, TR. I exclude English language searches. Column (2) adds the interaction with local population share by country of origin (linear term for population share is also included). Column (3) excludes additional search locations (BE, BG, DK, FI, NL, NO) to safeguard against correlated exposures. Standard errors clustered at the level of origin countries in parentheses, *** p<0.01, ** p<0.05, * p<0.1.

Table 5 shows a strong spillover of interest from origins to same-language speakers abroad. Column (2), in addition, shows that the larger is the share of language o speakers in country c, the stronger is the effect. Finally, column (3) restricts the sample further, excluding searches coming from "intermediately" exposed countries (BE, BG, DK, FI, NL, and NO), and still delivers a very strong and significant positive estimate for attention spillover. Thus, even those living in relatively non-affected places receive significant spillovers of interest in the Refugee Crisis from the origins.

Finally, I estimate dynamic β_t coefficients to test if interest in the topic "Refugees" from outside of the linguistic origins follows that at the origins especially closely when the issue is most salient. As one can see from Figure 11, coefficients are largely not significant and do not display any stable pattern before the Refugee Crisis begins. However, once the Refugee Crisis intensifies in 2015-2016, interest in this topic from abroad follows closely the interest at the linguistic origin. Spillovers become much less clear in 2018-2019.



Figure 11: Dynamic estimates of search intensity spillovers from linguistic origins to abroad.

4.2 Network homophily: learning from like-minded groups

Who do immigrants receive opinions from? And why do those who left respond to opinion change among those who stayed behind? In this section, I show that immigrants are particularly responsive to changes among the like-minded groups at the origins.

The Refugee Crisis was a polarizing event not only across countries, but within many countries as well. Using the 0 to 10 political left-right scale from the ESS, I divide respondents into 3 roughly equally sized categories, "Left", "Center", and "Right", and calculate conditional averages at the *origin* \times *group* \times *time* level. As illustrated on Figure 12, the Refugee Crisis induced a divergence between Left- and Right-leaning respondents. I exploit this *origin* \times *group* \times *time* variation in political attitudes to demonstrate the role of network homophily in learning from the origins.

The idea of network homophily would suggest that communications and opinion exchanges are more intense among the like-minded groups. And indeed, Figure B13 shows



The Refugee Crisis began between rounds 6 and 7 (marked by the vertical line). Left, Right, and Center political groups are based on the 0 to 10 left-right scale from the ESS, where Left is composed of those with scores from 1 to 4, Center is composed of those with 5, and Right is composed of those with scores from 6 to 10. Each group's size is roughly 1/3 across all rounds.

Figure 12: Dynamics of opposition towards non-European immigrants, by political stance.

that immigrants identifying as politically left-leaning demonstrate a very strong and significant co-movement with left-leaning peers at the origins, but do not co-move with other political subgroups there. An even stronger evidence for homophily comes from separating respondents into left and "non-left" political subgroups. In that case, the homophily effects for non-left are stronger than when I have Center and Right groups separately. To test more rigorously whether immigrants respond particularly strongly to changing opinions among the like-minded groups at the origin, I estimate the following model:

$$Y_{i,o,g,c,t} = \alpha + \beta \cdot Y_{o,g,t}^{Orig.Av} + \eta X'_{i,o,g,c,t} + \phi_o + \tau_{c,t} + \psi_{o,g} + \kappa_{g,t} + \varepsilon_{i,c,t,r,g}$$
(6)

where $Y_{o,g,t}^{Orig.Av}$ stands for average attitudes among group g in country of origin o in period t. Importantly, I identify β controlling for $\kappa_{g,t}$, the overall dynamics of group g, and for $\psi_{o,g}$, which captures any specific characteristics of political spectrum by country of origin. Moreover, I also estimate β_t coefficients separately for each time period, to show that homophily effects in learning from the origins activate when the Refugee Crisis is most salient.

Table 6 reports the estimates of β from model (6). In all specifications, the origin-country average among the politically like-minded group exerts a strong effect on attitudes of the

	(1)	(2)	(3)	(4)	(5)			
VARIABLES	Oppose non-European immigration							
Oppose non-Eur. immigr. (origin group)	0.262***	0.151**	0.328**	0.313*	0.384**			
	(0.070)	(0.067)	(0.156)	(0.162)	(0.175)			
Oppose non-Eur. immigr. (origin)			-0.177	-0.208				
			(0.165)	(0.170)				
Observations	18,527	18,527	18,527	18,527	18,527			
Adjusted R-squared	0.175	0.176	0.176	0.176	0.175			
Standard FEs	Yes	Yes	Yes	Yes	Yes			
Individual controls	Yes	Yes	Yes	Yes	Yes			
Group x round FE	Yes	Yes	Yes	Yes	Yes			
Group x origin FEs	No	Yes	Yes	Yes	Yes			
Origin time trends	No	No	No	Yes	Yes			
Origin x round FEs	No	No	No	No	Yes			

Table 6: Origin country spillovers from similar groups

Outcome variable is an individual-level response to a question "to what extent do you think [country] should allow people form the poorer countries outside of Europe?", with answers ranging from 1 (allow many) to 4 (allow none). Main treatment variable is the country-round-subgroup average value for the same question asked in respondent's country of origin. Subgroups are defined by political ideology: Left vs. Center & Right. Columns (3) and (4) control for simple origin-country average attitudes. Standard errors, clustered at the level of origin countries, are in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

expats. Moreover, as one can see from columns (3) and (4), simple origin-country average no longer matters once I include the subgroup origin average. Thus, this evidence suggests that most of learning from the origins is actually learning from like-minded origin-country subgroups. In column (5), I include a full set of Fixed Effects, including the Origin \times Round FEs, so the effects of changes in origin-country subgroups are identified out of deviations from origin-country averages in each period (see Figure 12).

These results suggest that because of the network homophily effect, immigrants learn disproportionately from opinions prevalent among like-minded groups at the origins. Moreover, as one can see from Figure 13, the extent of learning from like-minded peers at the origins intensifies when the Refugee Crisis is most salient. In addition, origin-country effects are stronger when immigrants are not themselves exposed to the Crisis (host countries are neither major asylum nor transit countries, Figure B14).

These results on the importance of network homophily align very well with the evidence from Section 4.3 on the importance of Internet and social media ties to the origins. As was documented in several recent papers, social media often supplies like-minded news and opinions, and contributes to increased political polarization, Halberstam and Knight (2016), Allcott et al. (2020), Levy (2021) and a review in Zhuravskaya et al. (2020). Thus, the fact that most of the origin-country spillover comes from like-minded groups, together with



Figure 13: Dynamic spillover coefficients from like-minded groups at the origins.

the fact that spillovers are much stronger for regions and individuals well-connected to the origins via social media, suggest that social media is likely to be the key mode of transmission of attitudes from origins to immigrants. Moreover, this section suggests that while shared identity, culture and worldviews among co-ethnics living in different countries may be part of the story behind the co-movement in attitudes and policy preferences, the actual communication over networks is an important separate mechanism.

4.3 Integration at destination and network ties to the origins

How strongly immigrants respond to origin-country news and opinions may depend on their attachment to destination and ties to the origins. Below I test if this is indeed the case for several political topics explored in Section 2.

4.3.1 Role of immigrants' integration at destination

Beginning with the opposition towards non-Europeans as the main outcome, Figure 14 reports the estimates of interaction coefficients. The main result is that better integration into the host society weakens co-movements with the origins. Naturalization significantly decreases co-movement coefficient³⁵, while speaking origin language at home strengthens comovements with the origins. Slightly less significantly, longer residence in the host country weakens the co-movement. Immigrants with tertiary education and those interested in politics tend to receive stronger spillovers from the origins, potentially because they pay more attention to salient political events happening abroad.



Figure 14: Individual-level interactions: role of immigrants' integration and other factors

To show that immigrants' social integration matters across a broader set of topics, on the Appendix Figure B16 I report similar interaction coefficients for the six political attitudes showing strong co-movement patterns in Section 2. The main result remains the same: better integration into the host society weakens immigrants' co-movement with the origins.

4.3.2 Network ties to the origins: the diaspora, the Internet, the family

How do information and attitudes get from origins to immigrants? In this section, I find suggestive evidence for three network channels: (i) local diasporas, (ii) social media connections to the origins, and (iii) cross-border family ties.

³⁵In the Appendix, Figure B15 shows that the relative strength of co-movement with host- and origincountry attitudes depends on the citizenship status. Naturalized immigrants show much stronger comovement with destination, and a much weaker one with the origins.

Co-national diasporas in the host regions

To measure sub-national (NUTS-2 level) population shares by country of origin, I use data from Alesina et al. (2021). While available only for 16 European countries, this data allows me to test whether living in a region with a larger co-national diaspora makes one more responsive to origin-country changes.

Panel (a) of Figure 15 shows that larger local diasporas correspond to stronger spillovers of political preferences from the origins. For example, looking at the support towards the EU integration, each additional percent of co-nationals in the local population increases the spillover coefficient by 0.03 units.



(a) Share of co-nationals

(b) Share of Facebook ties at the origins

Figure 15: Interaction coefficients with (a) regional size of the local diaspora, and (b) regional Facebook ties to the origins. Several salient topics.

To take into account potential selection into regions with more co-nationals, I calculate the ratio of regional share of co-nationals to the host-country share of co-nationals, and split the sample into two parts based on the median (around 1) of this variable. As one can see on Figure 16, the results get even sharper: spillovers from the origins are almost absent for respondents living in regions where co-nationals are under-represented as compared to the host country overall. This can be the result of two forces. First, smaller diasporas provide less information and less social pressure to comply with origin-country norms. Second, people may select to regions with larger diasporas based on their level of attachment to the origins and integration at destination³⁶.

³⁶Indeed, immigrants speaking origin language at home, less educated, and those who migrated recently - all are more likely to live in regions with large diasporas.



Figure 16: Binscatter plot: co-movement with the origins for immigrants living in regions with larger (left) and smaller (right) co-ethnic diasporas

Social media ties to the origins

To measure online social media ties to the origins I use data from Facebook's Social Connectedness Index (SCI) introduced in Bailey et al. (2018). The SCI is a measure of connectivity between a pair of geographic regions (i, j) based on the number of Facebook friendship links as a share of maximum possible number of links. Formally, $SCI_{i,j} = \frac{Connections_{i,j}}{Users_i \cdot Users_j}$ ³⁷. Panel (b) on Figure 15 shows that the larger is the share of region *i* Facebook friendships that go to origin-country *o*, the stronger is the spillover coefficient for immigrants from that origin living in that region. As before, this holds for several salient topics.

Quite naturally, the size of the local diaspora and the share of Facebook ties to the origins are very closely correlated, see Figure B18 in the Appendix³⁸, which makes it harder to separate the two effects. To side-step this issue, I measure the intensity of local interactions and the intensity of Internet use.

First, I use the ESS question about the frequency of meetings with friends, colleagues, and relatives. In the Appendix Table C6, columns (1)-(3), I show that the effect of large diasporas is only there for respondents who frequently meet with their friends and colleagues. Second, I use the ESS variable that gives the frequency of using the Internet. Columns (4)-(5) of Table C6 show that respondents not socially active locally receive significant spillovers from the origins only if they actively use the Internet. In contrast, the Internet does not

³⁷Figure B17 shows ties to Germany at the NUTS-2 level, confirming a significant variation at the regional level, and that geographically/linguistically closer regions have stronger social media ties to Germany.

 $^{^{38}}$ A 1 p.p. higher local share of population from country *i* corresponds to a 1.28 p.p higher share of local social media ties with country *i*. Thus, immigrants' tend to have social media ties disproportionately in their countries of origin.

play any role for locally active respondents. Thus, there is suggestive evidence that both local and online communications facilitate spillovers from the origins³⁹.

Cross-border family ties

To examine the role of family networks in cross-border transmission of attitudes, I use the data on local household composition from the ESS. This allows me to distinguish between immigrants who live together with their parents and those who do not. Especially for 1st-gen immigrants, not having parents in the household usually means that they remain at the origins, creating an additional channel of transmission from the origins. And indeed, columns (1) and (2) in Table C7 show that co-movement with the origins is weaker for immigrants living with their parents⁴⁰, more so for 1st-gen immigrants.

Additional evidence supporting the within-family transmission comes from the fact that higher parental education strengthens the co-movement with the origins as much as does immigrants' own education, see columns (3)-(5) in Table C7. Assuming that a higher parental educational corresponds to a stronger interest in politics, a positive interaction with parental education suggests that parental opinions constitute an important transmission channel.

5 Extensions and generalizations

In this section I provide evidence for the external validity of the "learning from the origins" phenomenon. First, I show that European Refugee Crisis affects political preferences of 1st- and 2nd-generation immigrants in the US. Second, I show that the passage of same-sex marriage laws at the origins also sends spillover effects to the diasporas abroad.

5.1 Cross-Atlantic spillovers: European Refugee Crisis in the US

While the core of the paper focuses on intra-European spillovers, this section documents political spillovers from Europe to the US. Using the General Social Survey (GSS) data, I show that when European countries accept and host refugees after 2015, US respondents with

³⁹Traditional media, such as TV, can also transmit opinions from the origins. However, since the majority of TV channels target local audiences, immigrants are predominantly exposed to host-country opinions via TV. Using the ESS question on the amount of time spent on political news via TV, Figure B19 shows that an hour of TV weakens the co-movement by 0.06 units (around 20% of the baseline coefficient).

⁴⁰I control for the interaction of origin-country attitudes with respondents' age, because age and cohabitation with parents are strongly correlated. Moreover, I may underestimate the true within-family spillovers, as those living separately from parents are probably less attached to their family and origins.

family origins in these countries show improved attitudes towards immigrants and become more political liberal overall. One important advantage of looking at spillovers across the Atlantic is that the US is far away from the Crisis's epicenter alleviating further any concerns about correlated exposures to the Crisis. Moreover, the GSS allows identifying immigrants down to 4th generation, so it's possible to test how the strength of reaction to origin-country events fades with each additional generation.

	(1)	(2)	(3)	(4)	(5)
VARIABLES	Oj	opose immig	ration to the	US	Conservative
Origin treated (Host Refs.) x Post2016	-0.535***	-0.507***	-0.182**	-0.072	-0.307*
	(0.150)	(0.143)	(0.077)	(0.044)	(0.153)
Observations	678	678	1,816	5,215	1,005
Adjusted R-squared	0.070	0.063	0.060	0.073	0.066
Origin FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
Host Region FE	Yes	Yes	Yes	Yes	Yes
Host Region x Time FEs	No	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes
Sample	1-2 gen	1-2 gen	1-2-3 gen	1-2-3-4 gen	1-2 gen

Table 7: DID estimates of origin-country asylum treatment on immigrants in the US.

Outcome variable is an individual-level response to a question "Do you think the number of immigrants to America nowadays should be...?", with answers ranging from 1 (increased a lot) to 5 (reduced a lot). Main treatment variable is the origin-country treatment status interacted with the post-2016 dummy (the first survey year after the Refugee Crisis began in Europe). Columns (1)-(2) only use 1st and 2nd gen migrants. Column (3) adds 3rd gen migrants, and column (4) adds 4th gen migrants. Column (5) uses a measure of liberal vs. conservative values, on a scale from 1 (extremely liberal) to 7 (extremely conservative). Individual controls include gender, age, age squared, marital status, employment status, education, and race. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

The GSS rounds are conducted every two years, and I focus on years 2006-2018, for which the GSS consistently recorded respondents' policy preferences on whether the number of immigrants in the US should be increased or decreased⁴¹. The first post-Crisis round is 2016. Column (1) in Table 7 shows that when respondents' country of origin accepts and hosts refugees in 2015-2016 (e.g., Germany or Sweden) they reduce their opposition to additional immigration to the US, as compared to respondents coming from European countries that did not host many refugees (such as Hungary, Great Britain, or Spain). The effect is not driven by differential exposure to local shocks in the US, as suggested by column (2), and is quantitatively large: about 1/2 of the pre-treatment s.d. of 1.1. There is a clear

⁴¹Unfortunately, unlike the ESS, the GSS does not allow distinguishing between various types of immigrants. Moreover, questions on opinions about refugees are asked in only one round in the GSS.



Figure 17: Event study: the effect of Crisis at the origins (asylum treatment) on the opposition towards immigration in the US

weakening of reaction to origin-country events with each subsequent generation of ancestry: column (3) adds 3rd-gen immigrants (defined as in Algan and Cahuc (2010)) and shows a much weaker effect. Column (4) adds 4th-gen immigrants, and the effect almost disappears. Finally, column (5) looks at the self-reported liberal vs. conservative stance and shows that respondents from Asylum-treated origins become more liberal overall.

Figure 17 shows the dynamic DID estimates, and confirms (i) the absence of significant pre-trends across groups of immigrants in the US, and (ii) that the effect of the Crisis kicks in 2016, and remains strong until at least 2018⁴².

5.2 Same-sex marriage laws and LGBT attitudes abroad

Does the "learning from the origins" phenomenon generalize to other events and topics, not directly related to migration? Below I use the staggered passage of same-sex marriage (SSM) laws across European countries and the ESS data on opposition to LGBT rights to test whether 1st- and 2nd-gen immigrants abroad are affected by the SSM laws at the origins. While each individual SSM law was less salient in the media and the public discourse than the Refugee Crisis of 2015, the staggered passage of these laws allows for a better identification within the event-study framework. Figure B20 in the Appendix shows when a SSM law passed in each country.

⁴²If I increase the sample size by either (i) adding respondents of non-European origins to the control group, or (ii) adding 3rd-generation immigrants, the absence of pre-trends becomes much clearer.

Beginning with a simple DID model, columns (1)-(2) in Table 8 show that opposition towards the LGBT among the expats abroad weakens in the years following the passage of the SSM law at the origins. Columns (3)-(4) show that, as before, the effect is driven by non-citizens at destination and 1st-gen immigrants. Importantly, liberal reforms at the origins create divergent reactions across the political spectrum abroad: column (5) reveals that left-leaning respondents increase their support of the LGBT freedoms, while center and right-leaning respondents do not conform to the policy change at the origins. Column (6) shows that these results hold and are even more pronounced for immigrants living in countries that never legalized same-sex marriage locally, potentially because local influence does not saturate origin-country effects in such host countries.

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	(1)	(=)	Oppose L	(0)		
After SSM law passed (origin)	-0.108***	-0.075**	-0.184***	-0.130***	-0.136**	-0.191***
	(0.035)	(0.032)	(0.045)	(0.035)	(0.053)	(0.066)
After SSM law passed (origin) x Citizen			0.174***			
			(0.043)			
After SSM law passed (origin) x 2 nd gen				0.163***		
				(0.030)		
After SSM law passed (origin) x Center					0.043	0.106*
					(0.037)	(0.054)
After SSM law passed (origin) x Right					0.119*	0.374***
					(0.068)	(0.117)
Observations	34,350	34,350	34,317	34,350	29,342	15,055
Adjusted R-squared	0.271	0.275	0.277	0.278	0.279	0.279
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes
Host FE	Yes	Yes	Yes	Yes	Yes	Yes
Host x round FE	No	Yes	Yes	Yes	Yes	Yes
Region FE	Yes	Yes	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes	Yes
Group x round FE	No	No	No	No	Yes	Yes
Sample	Full	Full	Full	Full	Full	Host never
						SSM law

Table 8: Same-sex marriage (SSM) laws at the origin and LGBT attitudes abroad

Outcome variable is an individual-level response to a question "Please say to what extent you agree or disagree with each of the following statements: Gay men and lesbians should be free to live their own life as they wish", with answers ranging from 1 (agree strongly) to 5 (disagree strongly). Main treatment variable is the origin-country passage of the same-sex marriage legislation, with values of 1 for time periods after the passage of the legislation. Column (3) interacts the main treatment with the host-country citizenship status. Column (4) interacts with the generation of migration (1st or 2nd). Column (5) interacts the main treatment with Left-Center-Right political identification of the respondents (smaller sample size). Column (6) retains only those expats currently residing in host countries that never passed a SSM law themselves. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

Exploring the homophily effect and the dynamics of the opinion response abroad further, on Figure 18, I show the event-study estimates for respondents on political Left and political Right. Clearly, there is no evidence for pre-trends among either of the two groups. There is a strong positive response to origin-country SSM law among Left-leaning expats, but a backlash from the Right-leaning expats. Both effects, however, are short-lived, potentially because the event salience is smaller and shorter than in the Refugee Crisis case.



Figure 18: Dynamics of opposition to LGBT: event-study around SSM laws at the origins.

6 Conclusion

This paper documented the phenomenon of "learning from the origins" in the dynamics of political preferences and voting behaviors: following salient events, political opinions spill over from abroad via pre-existing migration networks. There are significant spillovers from origins to (1st and 2nd-gen) immigrants for a range of topics: opposition to non-Europeans, anti-LGBT attitudes, and so on. Three channels facilitate the transmission: (i) local conational diasporas, (ii) online social ties to the origins, and (iii) within-family cross-border networks. Social integration at destination, such as naturalization, tends to weaken contagion from abroad, underscoring another reason for the importance of immigrants' integration.

To get closer to causality, this paper focused on the European Refugee Crisis of 2015, one of last decade's most salient events that strongly affected some European countries, but did not affect others. Origin-country exposure to the Refugee Crisis and associated asylum policies can be seen as plausibly exogenous for the expats, which allows evaluating their effect on (i) opposition to non-Europeans, and (ii) support for far-right parties among the diasporas abroad. The main finding is that favorable asylum policies at the origins (accepting and hosting refugees) reduced opposition to non-Europeans and far-right voting among the diasporas abroad. In contrast, large transitory refugee inflows through the origins sent abroad the backlash, similar to what was happening in the main transition countries.

Evidence from online search activity suggests that salient events attract disproportionate attention of co-nationals from abroad. Moreover, evidence on the importance of network homophily suggests that most of the learning from the origins is driven by like-minded groups. Thus, in the era of online media, when maintaining attention to distant friends and events in real-time is virtually cost-less, there emerges a fertile ground for the cross-border spread of ideology, both populist/xenophobic and more liberal. These results generalize beyond refugee attitudes (similar spillovers following the passage of same-sex marriage laws) and to non-European settings (Refugee Crisis spillovers are felt in the US).

Further research can shed more light on the mechanisms behind the international contagion of political preferences. How much of these effects are explained by inter-personal transmission vs. media consumption? Is preference/belief updating mostly driven by identity and social pressure or by rational learning? In the age of modern ICTs, do we learn from distant places as much as we do from events nearby? These and related questions may enrich the research on cultural and political change, polarization, and immigrants' integration.

References

- Agostinelli, Francesco, Matthias Doepke, Giuseppe Sorrenti, and Fabrizio Zilibotti, "It Takes a Village: The Economics of Parenting with Neighborhood and Peer Effects," Working Paper 27050, National Bureau of Economic Research April 2020.
- Ahern, Kenneth R., Ran Duchin, and Tyler Shumway, "Peer Effects in Risk Aversion and Trust," *Review of Financial Studies*, 2014, 27 (11), 3213–3240.
- Ajzenman, Nicolás, Cevat Giray Aksoy, and Sergei Guriev, "Exposure to transit migration: Public attitudes and entrepreneurship," *Journal of Development Economics*, 2022, 158, 102899.
- Akay, Alpaslan, Oliver Bargain, and Klaus Zimmermann, "Home Sweet Home? Macroeconomic Conditions in Home Countries and the Well-Being of Migrants," *Journal of Human Resources*, 2017, 52 (2), 351–373.
- Akerlof, GA and RE Kranton, "Economics and identity," QUARTERLY JOURNAL OF ECONOMICS, AUG 2000, 115 (3), 715–753.

- Aksoy, Cevat G., Christopher S. Carpenter, Ralph De Haas, and Kevin D. Tran,
 "Do laws shape attitudes? Evidence from same-sex relationship recognition policies in Europe," *European Economic Review*, 2020, 124, 103399.
- Albert, Christoph and Joan Monras, "Immigration and Spatial Equilibrium: The Role of Expenditures in the Country of Origin," *American Economic Review*, November 2022, 112 (11), 3763–3802.
- Alesina, Alberto and Marco Tabellini, "The Political Effects of Immigration: Culture or Economics?," *Harvard Business School Working Paper*, 2021, 79 (21-069).
- _, Armando Miano, and Stefanie Stantcheva, "Immigration and Redistribution," Working Paper 24733, National Bureau of Economic Research June 2018.
- _, Elie Murard, and Hillel Rapoport, "Immigration and preferences for redistribution in Europe," Journal of Economic Geography, 2021, 21 (6), 925–954.
- Algan, Yann and Pierre Cahuc, "Inherited Trust and Growth," American Economic Review, 2010, 100 (5), 2060–2092.
- Allcott, Hunt, Luca Braghieri, Sarah Eichmeyer, and Matthew Gentzkow, "The Welfare Effects of Social Media," American Economic Review, March 2020, 110 (3), 629– 76.
- Allport, Gordon, The nature of prejudice, Addison-Wesley Publishing Company, 1954.
- Altındağ, Onur and Neeraj Kaushal, "Do refugees impact voting behavior in the host country? Evidence from Syrian refugee inflows to Turkey," *Public Choice*, 2021, 186 (1), 149–178.
- Atkin, David, "The Caloric Costs of Culture: Evidence from Indian Migrants," American Economic Review, 2016, 106 (4), 1144–1181.
- Autor, David, David Dorn, Gordon Hanson, and Kaveh Majlesi, "Importing Political Polarization? The Electoral Consequences of Rising Trade Exposure," American Economic Review, October 2020, 110 (10), 3139–83.
- Bailey, Michael, Ruiqing Cao, Theresa Kuchler, and Johannes Stroebel, "The Economic Effects of Social Networks: Evidence from the Housing Market," *Journal of Political Economy*, 2018, 126 (6), 2224–2276.
- Barsbai, Toman, Hillel Rapoport, Andreas Steinmayr, and Christoph Trebesch,
 "The Effect of Labor Migration on the Diffusion of Democracy: Evidence from a Former Soviet Republic," American Economic Journal: Applied Economics, 2017, 9 (3), 36–69.
- Beach, Brian and W Walker Hanlon, "Culture and the Historical Fertility Transition," The Review of Economic Studies, 08 2022. rdac059.

- Bentzen, Sinding Jeanet, "Acts of God? Religiosity and Natural Disasters Across Subnational World Districts^{*}," *The Economic Journal*, 05 2019, *129* (622), 2295–2321.
- Bisin, Alberto and Thierry Verdier, ""Beyond the Melting Pot": Cultural Transmission, Marriage, and the Evolution of Ethnic and Religious Traits," *Quarterly Journal of Economics*, 2000, 115 (3), 955–988.
- and _ , "The Economics of Cultural Transmission and the Dynamics of Preferences," Journal of Economic Theory, 2001, 97 (2), 298–319.
- Bordalo, Pedro, Nicola Gennaioli, and Andrei Shleifer, "Salience Theory of Choice Under Risk," *The Quarterly Journal of Economics*, 06 2012, *127* (3), 1243–1285.
- Borjas, George J., "Ethnic Capital and Intergenerational Mobility"," The Quarterly Journal of Economics, 02 1992, 107 (1), 123–150.
- Bratti, Massimiliano, Claudio Deiana, Enkelejda Havari, Gianluca Mazzarella, and Elena Claudia Meroni, "Geographical proximity to refugee reception centres and voting," *Journal of Urban Economics*, 2020, *120*, 103290.
- Bredtmann, Julia, "Immigration and electoral outcomes: Evidence from the 2015 refugee inflow to Germany," *Regional Science and Urban Economics*, 2022, *96*, 103807.
- Bronnenberg, Bart, Jean-Pierre H. Dubé, and Matthew Gentzkow, "The Evolution of Brand Preferences: Evidence from Consumer Migration," *American Economic Review*, 2012, 102 (6), 2472–2508.
- Burchardi, Konrad B., Thomas Chaney, and Tarek A. Hassan, "Migrants, Ancestors, and Foreign Investments," *The Review of Economic Studies*, 2019, *86* (4), 1448–1486.
- Bursztyn, Leonardo and Davide Cantoni, "A Tear in the Iron Curtain: The Impact of Western Television on Consumption Behavior," *The Review of Economics and Statistics*, 03 2016, 98 (1), 25–41.
- _, Florian Ederer, Bruno Ferman, and Noam Yuchtman, "Understanding Mechanisms Underlying Peer Effects: Evidence From a Field Experiment on Financial Decisions," *Econometrica*, 2014, 82 (4), 1273–1301.
- Dal Bo, Ernesto, Frederico Finan, Olle Folke, Torsten Persson, and Johanna Rickne, "Economic Losers and Political Winners: Sweden's Radical Right," 2019, p. 44.
- DellaVigna, Stefano, Ruben Enikolopov, Vera Mironova, Maria Petrova, and Ekaterina Zhuravskaya, "Cross-Border Media and Nationalism: Evidence from Serbian Radio in Croatia," American Economic Journal: Applied Economics, 2014, 6 (3), 103–132.

- Dohmen, Thomas, Armin Falk, David Huffman, and Uwe Sunde, "The Intergenerational Transmission of Risk and Trust Attitudes," *The Review of Economic Studies*, 2012, 79 (2), 645–677.
- **Dustmann, Christian, Kristine Vasiljeva, and Anna Piil Damm**, "Refugee Migration and Electoral Outcomes," *The Review of Economic Studies*, 2019, *86* (5), 2035–2091.
- Fernández, Raquel and Alessandra Fogli, "Fertility: The Role of Culture and Family Experience," *Journal of the European Economic Association*, 2006, (4), 552–561.
- and _, "Culture: An Empirical Investigation of Beliefs, Work, and Fertility," American Economic Journal: Macroeconomics, January 2009, 1 (1), 146–77.
- Fouka, Vasiliki and Hans-Joachim Voth, "Collective Remembrance and Private Choice: German-Greek Conflict and Behavior in Times of Crisis," *Mimeo*, 2021.
- Galor, Oded and Omer Moav, "Natural Selection and the Origin of Economic Growth*," The Quarterly Journal of Economics, 11 2002, 117 (4), 1133–1191.
- and Ömer Özak, "The Agricultural Origins of Time Preference," American Economic Review, 2016, 106 (10), 3064–3103.
- Giavazzi, Francesco, Ivan Petkov, and Fabio Schiantarelli, "Culture: persistence and evolution," *Journal of Economic Growth*, 2019, 24, 117–154.
- Giuliano, Paola, "Living Arrangements in Western Europe: Does Cultural Origin Matter?," Journal of the European Economic Association, 2007, p. 26.
- and Marco Tabellini, "The Seeds of Ideology: Historical Immigration and Political Preferences in the United States," NBER Working Paper 27238, 2020.
- and Nathan Nunn, "Understanding Cultural Persistence and Change," The Review of Economic Studies, 12 2020, 88 (4), 1541–1581.
- Golub, Benjamin and Matthew O. Jackson, "How Homophily Affects the Speed of Learning and Best-Response Dynamics," *The Quarterly Journal of Economics*, 07 2012, 127 (3), 1287–1338.
- **Guriev, Sergei and Elias Papaioannou**, "The Political Economy of Populism," *Journal of Economic Literature*, September 2022, *60* (3), 753–832.
- Halberstam, Yosh and Brian Knight, "Homophily, group size, and the diffusion of political information in social networks: Evidence from Twitter," *Journal of Public Economics*, 2016, 143, 73–88.
- Hangartner, Dominik, Elias Dinas, Moritz Marbach, Konstantinos Matakos, and Dimitrios Xefteris, "Does Exposure to the Refugee Crisis Make Natives More Hostile?," American Political Science Review, 2019, 113 (2), 442–455.

- Koch, Caleb M., Izabela Moise, Dirk Helbing, and Karsten Donnay, "Public debate in the media matters: evidence from the European refugee crisis," *EPJ Data Sci.*, 2020, 9 (1), 12.
- Lebow, Jeremy, Jonathan Moreno-Medina, Salma Mousa, and Horacio Coral, "Fertility: The Role of Culture and Family Experience," *Mimeo*, 2023.
- Levy, Ro'ee, "Social Media, News Consumption, and Polarization: Evidence from a Field Experiment," American Economic Review, 2021, 111 (3), 831–70.
- Luttmer, Erzo F. P and Monica Singhal, "Culture, Context, and the Taste for Redistribution," American Economic Journal: Economic Policy, 2011, 3 (1), 157–179.
- Manski, Charles F., "Identification of Endogenous Social Effects: The Reflection Problem," *The Review of Economic Studies*, 1993, 60 (3), 531.
- Miho, Antonela, Alexandra Jarotschkin, and Ekaterina Zhuravskaya, "Diffusion of Gender Norms: Evidence from Stalin's Ethnic Deportations," SSRN Working Paper, 2020.
- Nekoei, Arash, "Immigrants' Labor Supply and Exchange Rate Volatility," American Economic Journal: Applied Economics, 2013, 5 (4), 144–64.
- Noury, Abdul and Gerard Roland, "Identity Politics and Populism in Europe," Annual Review of Political Science, 2020, 23 (1), 421–439.
- Ochsner, Christian and Felix Roesel, "Migrating Extremists," *The Economic Journal*, 03 2020, *130* (628), 1135–1172.
- and Felix Rösel, "Activated History The Case of the Turkish Sieges of Vienna," CESifo Working Paper No. 6586, 2017, p. 66.
- Rapoport, Hillel, Sulin Sardoschau, and Arthur Silve, "Migration and Cultural Change," *CESifo Working Paper No. 8547*, 2020, p. 57.
- Rooduijn, Matthijs, Stijn van Kessel, Caterina Froio, Andrea Pirro, Sarah De Lange, Daphne Halikiopoulou, Paul Lewis, Cas Mudde, and Paul Taggart, "The PopuList: an overview of populist, far right, far left and Eurosceptic parties in Europe," January 2019.
- Steinmayr, Andreas, "Contact versus Exposure: Refugee Presence and Voting for the Far-Right," *The Review of Economics and Statistics*, 2021, 103 (2), 310–327.
- Tian, Yuan, Maria Esther Caballero, and Brian Kovak, "Social Learning along International Migrant Networks," 2020.
- Zhuravskaya, Ekaterina, Maria Petrova, and Ruben Enikolopov, "Political Effects of the Internet and Social Media," Annual Review of Economics, 2020, 12 (1), 415–438.

Online Appendix

to

"Learning from the Origins"

A. Conceptual framework

In this section, I describe a simple conceptual framework to guide our thinking about how identity, information, and issue salience can shape the process of learning from the origins. The model is inspired by two strands of research: the economics of identity (Akerlof and Kranton (2000)) and the economics of salience and attention (Bordalo et al. (2012)).

Consider an individual who forms an opinion or decides on an action a regarding an issue A: for example, whether refugees coming to Europe bring more good than harm, or whether to vote for a far-right candidate. An individual derives utility from acting in a way aligned with the state of the world, a_t^* . However, an individual is also subject to a pressure to comply with the opinions/actions $a_{o,t}$ prevalent in group o that they identify with. Consider the following utility function that combines these two motives:

$$U_{i,o,t}(a_{i,o,t}, a_t^*, a_{o,t}) = \lambda \cdot (a_{i,o,t} - a_{o,t})^2 + (1 - \lambda) \cdot (a_{i,o,t} - a_t^*)^2,$$
(7)

where λ is the weight of identity concerns. State of the world is unknown, but individual *i* has a prior about a_t^* that is normally distributed around mean $a_{i,o}^{pr}$ with a variance $\sigma_{a^{pr}}^{2}$ ⁴³.

Attitudes and actions prevalent in the group one identifies with also carry a signal $s_{o,t}$ about the true state of the world. Assume for simplicity that individuals only pay attention to signals coming from groups they identify with, and that attitudes in group o are a noisy signal of the true state of the world, with precision $1/\sigma_{\varepsilon_{o,t}}^2$ and potentially some bias a_o , so $s_{o,t} = a_{o,t} = a_t^* + \varepsilon_{o,t}$, where $\varepsilon_{o,t} \sim \mathcal{N}(a_o, \sigma_{\varepsilon_{o,t}}^2)$. Individual i wants to take an action/form an opinion that solves

$$\max_{a_{i,o,t}} \lambda \cdot (a_{i,o,t} - a_{o,t})^2 + (1 - \lambda) \cdot \mathbb{E}((a_{i,o,t} - a^*)^2 | s_{o,t}).$$
(8)

It is straightforward to show that the solution to this problem is:

$$a_{i,o,t} = \lambda \cdot a_{o,t} + (1-\lambda) \cdot (\gamma_{o,t} \cdot a_{o,t} + (1-\gamma_{o,t}) \cdot a_{i,o}^{pr}), \tag{9}$$

where $\gamma_{o,t} = \frac{\sigma_{apr}^2}{\sigma_{apr}^2 + \sigma_{\varepsilon_{o,t}}^2}$ captures the relative precision of the signal versus the prior. Thus, more noisy information from group o (higher values of $\sigma_{\varepsilon_{o,t}}^2$) means an agent updates his or her own position less. Overall, equation (9) gives a micro-founded version of the linearin-means peer effects model, where peers' actions/opinions matter through two channels: identity and information. The extent to which individual i updates his own position $a_{i,o,t}$

⁴³Note that the prior can also depend on the group one belongs to: country of origin, or current country/region, or religious group, etc.

for each unit of change in the group he identifies with is given by:

$$\frac{\partial a_{i,o,t}}{\partial a_{o,t}} = \lambda + (1 - \lambda) \cdot \gamma_{o,t}.$$
(10)

Thus, a stronger sense of belonging to group o increases the extent of co-movement/adjustment to that group's actions and opinions. Moreover, more precise information coming from group o also increases the extent of adjustment to group's changing stance. Importantly, all the fixed biases and priors are naturally differenced out.

What determines the signal's precision? While in this simple version of the model signal's precision is exogenous, one can reasonably argue that it may depend on (i) the amount of information coming from a group (a larger number of noisy individual signals makes the average message more precise), (ii) the extent to which a group is an 'expert' in a given issue (certain groups may have more information about certain issues), (iii) issue salience among the members of a group, and so on. It is easy to extend this toy model to incorporate multiple sources of signals (e.g., adding host country attitudes as a source of information) and endogenous allocation of attention weights between host and origin societies.

B. Additional Figures



Figure B1: Opposition towards non-Europeans: among the expats and at the origins.



Figure B2: Opposition towards LGBT rights: among the expats and at the origins.



Opposition to non-Europeans

Figure B3: On the left are long-run differences between origins and immigrants. On the right are country-specific β coefficients from model (1).



Figure B4: Hosted refugee populations per capita (end of year), by country, and asylum application rejection rates (right axis). Both for non-European asylum seekers.



Figure B5: Increase in hosted refugees vs. pre-2015 opposition to non-Europeans



Figure B6: Changes in asylum rejection rates, by Asylum-treatment status.



Figure B7: Irregular border crossings, entry and transit countries. Based on the IOM data.



Figure B8: Refugee Crisis salience: % saying immigration is salient (Eurobarometer), share of refugees-related media articles (GDELT), and asylum applications (Eurostat)



Figure B9: Binscatter: opposition to non-Europeans and support for far-right parties among immigrants. Controls: origin FEs, destination x time FEs, and individual characteristics.



Figure B10: Yearly estimates of origin-country Transit treatment on voting for far-right parties by the expats abroad.



Figure B11: Increase in spillover coefficients after 2015: placebo topics.



Figure B12: GT search for Flüchtlinge (DE) and Refugiados (ES) coming from France.



Figure B13: Reactions of immigrants on the left to political subgroups at the origins



Figure B14: Dynamic spillover coefficients from like-minded groups at the origins, only non-treated host countries.



Figure B15: Allocation of attention between origin and destination: the role of naturalization



Figure B16: Interactions estimated jointly. Each figure focuses on a separate political topic. The estimates show how the co-movement coefficient changes with a given variable.



Figure B17: Facebook ties to Germany, NUTS-2 level subregions

···~ ø



Figure B18: Facebook ties to the origins and local population shares by country of origin



Figure B19: Interactions with news consumption via TV, basic and with controls.



Figure B20: The timing of same-sex marriage legislation across countries in Europe, together with the sampling of immigrants in the ESS rounds 1-9. NL (passed in 2000) are not shown.

C. Additional Tables

Summary Statistics, Immigra	nts and natives	
	Immigrants (1st	Natives
	and 2nd gen.)	
Tertiary Education	0.33	0.27
	(0.47)	(0.44)
Unemployed	0.051	0.040
	(0.22)	(0.20)
Second gen.	0.35	
-	(0.48)	
Citizen	0.69	
	(0.46)	
Length of Residence (years)	29.2	
0 0 /	(19.9)	
Oppose Non-Eur. Migration	0.48	0.53
	(0.50)	(0.50)
Political Left	0.35	0.31
	(0.48)	(0.46)
Political Right	0.30	0.36
5	(0.46)	(0.48)
N	38065	359308
Standard deviations in parenthes	205	

Table C1: Summary statistics, ESS sample: 1st and 2nd-gen immigrants vs. natives

Standard deviations in parentheses.

Immigrants' origin countrie	s	
Country of Origin	Number of	Share among all
	individuals	immigrants
Albania	597	0.014
Austria	711	0.017
Bosnia and Herzegovina	1499	0.036
Belgium	545	0.013
Bulgaria	469	0.011
Belarus	1132	0.027
Switzerland	299	0.007
Cyprus	164	0.003
Czech Republic	741	0.017
Germany	3284	0.079
Denmark	426	0.010
Estonia	291	0.007
Spain	692	0.016
Finland	807	0.019
France	1676	0.040
United Kingdom	2138	0.051
Greece	453	0.011
Croatia	740	0.017
Hungary	687	0.016
Ireland	555	0.013
Israel	161	0.003
Iceland	160	0.003
Italy	1942	0.047
Liechtenstein	10	0.0002
Lithuania	456	0.011
Luxembourg	155	0.003
Latvia	373	0.009
Moldova	330	0.007
Montenegro	33	0.0007
North Macedonia	235	0.005
Malta	136	0.003
Netherlands	693	0.016
Norway	365	0.008
Poland	2616	0.063
Portugal	1104	0.026
Romania	1676	0.040
Serbia	435	0.010
Russia	6945	0.168
Sweden	602	0.014
Slovenia	166	0.004
Slovakia	769	0.018
Turkey	1844	0.044
Ukraine	1979	0.047
Kosovo	214	0.005
Total	41305	1

Table C2: Origin countries of 1st and 2nd gen immigrants in the ESS sample

	(1)	(2)	(3)	(4)	(5)	(6)
	DID	DID	DID	DID	DID	DID
VARIABLES	Strict	Ref. not	Restrict	Oppose	Immigr.	Safety
	refugee	in fear of	family	income	worsen	important
	policy	persec.	reunif.	redistrib.	crime	concern
Origin Treated (Host Refs.) x Post2015	-0.124*	-0.095*	-0.107	-0.123***	-0.286**	-0.078**
	(0.068)	(0.053)	(0.071)	(0.033)	(0.110)	(0.034)
Observations	10,386	5,790	6,224	32,824	6,109	32,341
Adjusted R-squared	0.117	0.082	0.093	0.096	0.073	0.123
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes
Country x round FEs	Yes	Yes	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes	Yes
Sample	Full	Full	Full	Full	Full	Full

Table C3: DID estimates of origin-country asylum treatment: additional outcomes

Outcome variables (and hence sample sizes) differ across columns. Column (1): support for strict government policies towards granting refugee status. Column (2): support for the opinion that refugees are not in real fear of persecution in their own countries. Column (3): opposition to refugee family reunification. Column (4): opposition to government reducing differences in income levels. Column (5): opinion that immigrants worsen crime issues. Column (6): importance of personal safety concerns (average of two questions: on government providing safety, and on the importance of safe surroundings). Main treatment variable is the origin-country treatment status interacted with the post-2015 dummy. Treatment definition is based on the increase in hosted refugees per capita from 2013 to 2016. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

Table C	4: DID	estimates of	origin-country	asylum	treatment:	placebo	outcomes
				•/		1	

	(1)	(2)	(3)	(4)	(5)
	DID	DID	DID	DID	DID
VARIABLES	Trust local	Trust EU	Support	Oppose	General
	parl.	parl.	EU unif.	LGBT	trust
Origin Treated (Host Refs.) x Post2015	-0.073	0.100	0.129	-0.014	0.107***
	(0.058)	(0.111)	(0.148)	(0.037)	(0.039)
Observations	32,041	29,372	26,335	32,161	33,510
Adjusted R-squared	0.210	0.087	0.074	0.274	0.107
Origin FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
Country x round FEs	Yes	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes
Sample	Full	Full	Full	Full	Full

Outcome variables (and hence sample sizes) differ across columns. Column (1): trust in local parliament. Column (2): trust in the EU parliament. Column (3): support for the EU unification. Column (4): opposition to LGBT rights. Column (5): generalized trust. Main treatment variable is the origin-country treatment status interacted with the post-2015 dummy. Treatment definition is based on the increase in hosted refugees per capita from 2013 to 2016. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)			
VARIABLES	Oppose non-European immigration						
Rejection rate drop x Post2015	-0.303***	-0.307***	-0.230	-0.298***			
	(0.104)	(0.112)	(0.161)	(0.090)			
Observations	32,021	28,195	15,459	31,414			
Adjusted R-squared	0.155	0.159	0.126	0.155			
Origin FE	Yes	Yes	Yes	Yes			
Time FE	Yes	Yes	Yes	Yes			
Country x round FEs	Yes	Yes	Yes	Yes			
Individual controls	Yes	Yes	Yes	Yes			
Sample	Full	No transit	Host not	No recent			
		origins	treated	migrants			

Table C5: DID estimates of origin-country asylum rejection rates treatment

Outcome variable is an individual-level response to a question "to what extent do you think [country] should allow people form the poorer countries outside of Europe?" with answers ranging from 1 (allow many) to 4 (allow none). Main treatment variable is the origin-country drop in asylum rejection rates, displayed on Figure 5. Column (2) removes transit migration origins. Column (3) only retains the expats currently residing in host countries without significant exposure to the Crisis (asylum or transit). Column (4) omits immigrants who left their origins after 2012. Standard errors, clustered at the level of origin countries, are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(2)	(4)	(5)
		(2)	(3)	(4)	(5)
	FE	FE	_Fe	FE	FE
VARIABLES		Oppose n	on-Eur. 1mn	ngration	
Oppose non-Eur. migr. (origin)	0.012	0.167*	0.163	0.100	0.082
	(0.108)	(0.091)	(0.101)	(0.075)	(0.108)
Oppose non-Fur migr (origin) y Large Diasp	0.186**	0.029	0.028	(/	()
oppose non Eta: mgr. (origin) i Etage Etasp.	(0.076)	(0.056)	(0.026)		
Oppose non Fur migr (origin) y Most Social	(0.070)	(0.050)	0.110**		
Oppose non-Eur. migr. (origin) x Meet Social.			-0.119		
			(0.052)		
Oppose non-Eur. migr. (origin) x Meet Social. x L arge Diasp.			0.176**		
			(0.069)		
Oppose non-Eur. migr. (origin) x High Internet Usage				0.168**	0.039
				(0.075)	(0.110)
Observations	5,672	6,744	12,416	5,293	4,481
Adjusted R-squared	0.133	0.140	0.144	0.150	0.128
Origin FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
Host x Time FE	Yes	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes
Country-pair FE	Yes	Yes	Yes	Yes	Yes
Sample by local social activity	High	Low	Full	Low	High

Table C6: Transmission of opinions from the origins: local interactions vs. the Internet

Internet use variable is only available for rounds 1-5 and 8-9, so rounds 6 and 7 are omitted in columns (4) and (5), hence a smaller sample size. Cluster-robust standard errors in parentheses, *** p<0.01, ** p<0.05, * p<0.1.

	(1)	(2)	(3)	(4)	(5)
	FE	FE	FE	FE	FE
VARIABLES	Oppose non-Eur. immigration				
Oppose non-Eur. immigr. (origin)	0.279***	0.420***	0.098**	0.058	0.123**
	(0.089)	(0.104)	(0.038)	(0.037)	(0.047)
Oppose non-Eur. immigr. (origin) x Live with parents	-0.082*	-0.223**			
	(0.044)	(0.097)			
Oppose non-Eur. immigr. (origin) x Mother higher ed.			0.125**	0.088*	0.169***
			(0.056)	(0.047)	(0.040)
Oppose non-Eur. immigr. (origin) x Own higher ed.				0.115*	0.061*
				(0.062)	(0.032)
Observations	19,731	11,405	19,447	19,399	11,835
Adjusted R-squared	0.153	0.142	0.152	0.156	0.145
Origin FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
Country x round FEs	Yes	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes	Yes
Sample	Full	1 st gen	Full	Full	1 st gen

Table	C7:	Within-fami	ilv ties	and	co-movements	with	the	origins
Table	\bigcirc	vvioinin ionin	1 9 0100	and	co movemento	VV I UII	UIIC	originis

Outcome variable is an individual-level response to a question "to what extent do you think [country] should allow people form the poorer countries outside of Europe?", with answers ranging from 1 (allow many) to 4 (allow none). Main treatment variable is the country-round average value for the same question asked in respondent's country of origin. Columns (1)-(2) focus on interactions with whether immigrants live in the same household with their parents. Columns (3)-(5) focus on mother's education level (similar, but weaker, results are there for father's education). Both results are stronger for 1^{st} -gen immigrants (columns (2) and (5)). Standard errors clustered at the level of origin countries are in parentheses. *** p<0.01, ** p<0.05, * p<0.1.

D. Issue salience and political spillovers in the Eurobarometer

In this section, I augment and the main analysis of the paper with data from the Eurobarometer (EB). One advantage of the EB data is that its rounds are conducted at a much higher frequency than in the ESS: about 6 months between the rounds. Utilizing this finer temporal dimension, Table D1 reports basic results from estimating model (1) on the EB data, focusing on opposition towards non-EU immigration. Columns (1) to (5) and column (7) replicate some of the main specifications estimated on the ESS data in Table 1, and show very similar results, confirming the significance and strength of political spillovers from the origins. Quantitatively, the spillover coefficients are even stronger, potentially because of the finer time scale. Moreover, column (6) uses lagged attitudes at the origin to safeguard against the reverse direction of spillovers, showing very strong and significant results. Overall, it is reassuring to find similar results from a completely different dataset⁴⁴.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	FE	FE	FE	FE	FE	FE	FE
VARIABLES			Oppose n	on-EU imm	igration		
Oppose non-EU immigr. (origin)	0.476***	0.478***	0.458***	0.301***	0.298**		0.277**
	(0.094)	(0.096)	(0.106)	(0.105)	(0.128)		(0.132)
Oppose non-EU immigr. (local)			0.442*				
			(0.222)				
Oppose non-EU immigr. (origin), 1st lag						0.270**	
						(0.128)	
Observations	5,420	5,366	5,366	5,366	5,366	4,738	4,065
Adjusted R-squared	0.084	0.101	0.102	0.110	0.130	0.114	0.127
Origin FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Host Country FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Individual controls	No	Yes	Yes	Yes	Yes	Yes	Yes
Host x round FE	No	No	No	Yes	Yes	Yes	Yes
NUTS FE	No	No	No	No	Yes	No	No
Group x round FEs	No	No	No	No	No	No	Yes

Table D1: Eurobarometer: spillovers of opposition towards non-EU immigrants

Outcome variable is individual-level feeling towards immigration from outside of EU, ranging from 1 (very positive) to 4 (very negative). Main treatment variable is the country-round average value for the same question asked in respondent's country of origin. Column (3) adds local country-round averages. Column (6) estimates the effect of previous-round attitudes at the origins on current attitudes of emigrants. Column (7) allows for separate shocks for Left-Center-Right political groups (available for a subset of respondents), and for (non)tertiary educated groups of respondents. Standard errors, clustered at the level of origin countries, are in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

⁴⁴The downside of the EB data is that questions on the attitudes towards non-EU immigrants began to appear only from November 2014 onward - a time period already well into the Refugee Crisis. For this reason, it is hard to conduct a clear pre- and post-Crisis analysis in the spirit of Table 2. Moreover, the Eurobarometer does not allow to distinguish 1st- and 2nd-gen immigrants, which also constrains the analysis. An additional benefit of the EB is data on issue salience, which allows me to test whether changes in issue salience at the origins affect what immigrants perceive as salient, thereby contributing to the spread of opinions. The EB offers a list of economic and political issues (inflation, unemployment, health, education, immigration, crime, etc.) which respondents code as salient or not (i) for the country of residence, and (ii) the EU overall. Respondents can only code 1 or 2 issues as salient. I use data on immigration salience from 2011 to 2018 and estimate whether a higher immigration salience at the origins affects the perceptions of immigration salience among the expats (for the host country and for the EU overall).

To begin with, Figure D1 shows that there is a stronger increase in perceptions of local issue salience during and after the Refugee Crisis for the expats from more affected origins (either as asylum countries or as transit countries).



Figure D1: Local immigration salience among the expats, by origin-country treatment status

Moreover, Figure D2 reveals a strong co-movement in perceptions of broad (EU-level) immigration salience between immigrants and their origins. During the Refugee Crisis rounds (Nov 2014, May 2015, Nov 2015 and May 2016) the co-movement is the closest. This supports the key assumption of the paper: during the periods of salient events at the origins, topics deemed important at the the origins receive additional attention from the expats abroad⁴⁵.

⁴⁵Similar, but slightly weaker co-movements are observed for perceptions of local issue salience. Thus, for example, if Germans report immigration being salient for Germany, then Germans abroad tend to report immigration being salient for their current host country.



Figure D2: Co-movement of immigration salience (EU): immigrants and their origins

Could these patterns reflect correlated exposures to the Crisis or general trends in issue salience? Following the strategy used in models (1) and (5), I estimate spillover coefficients for issue salience, conditional on local dynamics and origin FEs. Moreover, to further address the issue of correlated exposure, I limit the sample of residence countries to those which were not strongly exposed to the Crisis. Figure D3 shows that among the diasporas living in nonaffected host countries, there is a growing perceptions of (EU-level) immigration salience when immigration becomes salient at the origins in 2014-2016. Figure D4 reveals somewhat weaker spillovers to local immigration salience (i.e., immigrants coming from countries where immigration becomes salient tend to increasingly report immigration being salient locally). Overall, this supports the attention and salience arguments made in section 4.1.



Figure D3: Spillovers of immigration salience: EU salience among the expats



Figure D4: Spillovers of immigration salience: local salience among the expats